SECURITIES AND EXCHANGE COMMISSION

FORM 8-K

Current report filing

Filing Date: **2007-10-23** | Period of Report: **2007-10-21** SEC Accession No. 0001299933-07-006122

(HTML Version on secdatabase.com)

FILER

ADC TELECOMMUNICATIONS INC

CIK:61478| IRS No.: 410743912 | State of Incorp.:MN | Fiscal Year End: 1031

Type: 8-K | Act: 34 | File No.: 000-01424 | Film No.: 071186290

SIC: 3661 Telephone & telegraph apparatus

Mailing Address EDEN PRAIRIE MN 55344

Business Address 13625 TECHNOLOGY DRIVE 13625 TECHNOLOGY DRIVE EDEN PRAIRIE MN 55344 9529388080

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of Earliest Event Reported): October 21, 2007

ADC Telecommunications, Inc.

(Exact name of registrant as specified in its charter)

Minnesota	0-1424	41-0743912
(State or other jurisdiction of incorporation)	(Commission File Number)	(I.R.S. Employer Identification No.)
13625 Technology Drive, Eden Prairie, Minnesota		55344
(Address of principal executive offices)		(Zip Code)
Registrant's telephone number, including area	a code:	952.938.8080
	Not Applicable	
Former name of	or former address, if changed since la	ast report
Check the appropriate box below if the Form 8-K filing the following provisions:	is intended to simultaneously satisfy	the filing obligation of the registrant under any o
[] Written communications pursuant to Rule 425 under [] Soliciting material pursuant to Rule 14a-12 under th [] Pre-commencement communications pursuant to Rule [] Pre-commencement communications pursuant to Rule []	e Exchange Act (17 CFR 240.14a-1: ale 14d-2(b) under the Exchange Act	2) (17 CFR 240.14d-2(b))

Top of the Form

Item 1.01 Entry into a Material Definitive Agreement.

Merger Agreement

On October 21, 2007, ADC Telecommunications, Inc., a Minnesota corporation ("ADC"), Hazeltine Merger Sub, Inc., a Delaware corporation and a wholly owned subsidiary of ADC ("Merger Sub"), and LGC Wireless, Inc., a Delaware corporation ("LGC"), entered into an Agreement and Plan of Merger (the "Merger Agreement"), providing for the acquisition of 100% of the outstanding shares of LGC. Following the merger of Merger Sub with and into LGC, LGC will continue as a wholly owned subsidiary of ADC.

The purchase price is approximately \$148.5 million, subject to adjustment if LGC does not have net assets of a predetermined level at the closing. In order to address potential indemnity claims of ADC, \$15.5 million of the purchase price will be held in escrow for up to 15 months following the closing and \$2.325 million of the purchase price will be held in escrow for up to 36 months following the closing. The remainder of the purchase price will be paid to equity holders of LGC at the closing. As partial payment of the purchase price, holders of inthe-money LGC options generally may elect either to receive cash for their options or to exchange their options for options to acquire shares of ADC. If all the option holders of LGC elect to exchange their options, then ADC would issue options worth approximately \$12.5 million. In addition to the purchase price, ADC also has agreed to assume certain transaction expenses, LGC employee bonuses and LGC's current debt, which aggregate to approximately \$20.5 million.

ADC and LGC have made customary representations, warranties and covenants in the Merger Agreement. The representations and warranties of each party set forth in the Merger Agreement have been made solely for the benefit of the other party to the agreement and such representations and warranties should not be relied on by any other person. In addition, such representations and warranties (i) have been qualified by disclosure schedules that the parties have exchanged in connection with the signing of the Merger Agreement, (ii) are subject to the materiality standards set forth in the Merger Agreement, which may differ from what may be viewed as material by investors, and (iii) were made only as of the date of the Merger Agreement or such other date as specified in the Merger Agreement. The disclosure schedules referred to above contain information that modifies, qualifies and creates exceptions to the representations and warranties set forth in the Merger Agreement. Accordingly, no person should rely on the representations and warranties as characterizations of the actual state of facts, as they are modified in important part by those disclosure schedules. Moreover, information concerning the subject matter of the representations and warranties may change after the date of execution of the Merger Agreement.

Consummation of the merger is subject to various customary conditions, including approval by the shareholders of LGC, the absence of legal impediment to the merger and the receipt of required regulatory approvals.

In connection with the Merger Agreement, the parties have also entered into certain other agreements, including employment agreements with certain executive officers of LGC. Certain shareholders of LGC have also entered into a voting agreement as described below.

A copy of the Merger Agreement is included herein as Exhibit 2.1 and is incorporated herein by reference. The foregoing description of the Merger Agreement is qualified in its entirety by reference to the full text of the Merger Agreement.

Voting Agreement

Concurrently with entering into the Merger Agreement, ADC entered into a voting agreement (the "Voting Agreement") with certain shareholders of LGC who collectively hold sufficient outstanding shares of LGC to approve the Merger (the "Shareholders"). The Voting Agreement provides that the Shareholders will vote their shares in favor of the Merger and against any other transaction presented to them that could have the effect of impeding the ability of ADC to consummate the Merger.

A copy of the Voting Agreement is included herein as Exhibit 10.1 and is incorporated herein by reference. The foregoing description of the Voting Agreement is qualified in its entirety by reference to the full text of the Voting Agreement.

Item 7.01 Regulation FD Disclosure.

On October 22, 2007, ADC issued a press release announcing the acquisition of LGC.

A copy of the press release is furnished herein as Exhibit 99.1.

Item 9.01 Financial Statements and Exhibits.

- (c) Exhibits.
- 2.1 Agreement and Plan of Merger, dated October 21, 2007, by and among ADC Telecommunications, Inc., Hazeltine Merger Sub, Inc. and LGC Wireless, Inc.
- 10.1 Voting Agreement, dated October 21, 2007, by and among ADC Telecommunications, Inc. and the Shareholders of LGC Wireless, Inc. listed on the signature page to the Voting Agreement
- 99.1 Press Release dated October 22, 2007

Top of the Form

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

ADC Telecommunications, Inc.

By: /s/ James G. Mathews

October 23, 2007

Name: James G. Mathews

Title: Vice President and Chief Financial Officer

Top of the Form

Exhibit Index

Exhibit No.	Description
2.1	Agreement and Plan of Merger
10.1	Voting Agreement
99.1	Press Release

AGREEMENT AND PLAN OF MERGER

by and among

ADC TELECOMMUNICATIONS, INC.,

HAZELTINE MERGER SUB, INC.

and

LGC WIRELESS, INC.

OCTOBER 21, 2007 AGREEMENT AND PLAN OF MERGER

This **AGREEMENT AND PLAN OF MERGER** (this "<u>Agreement</u>"), dated October 21, 2007, is made and entered into by and among ADC Telecommunications, Inc., a Minnesota corporation ("<u>Parent</u>"), Hazeltine Merger Sub, Inc., a Delaware corporation and wholly owned subsidiary of Parent ("<u>Merger Sub</u>"), and LGC Wireless, Inc., a Delaware corporation (the "<u>Company</u>"). Merger Sub and the Company are sometimes collectively referred to herein as the "<u>Constituent Corporations</u>".

WITNESSETH:

WHEREAS, the respective Boards of Directors of Parent, Merger Sub and the Company have determined that it is advisable and in the best interests of the respective corporations and their shareholders and stockholders that Merger Sub be merged with and into the Company in accordance with the Delaware General Corporation Law (the "<u>DGCL</u>") and the terms of this Agreement, pursuant to which the Company will be the surviving corporation and will be a wholly owned subsidiary of Parent (the "<u>Merger</u>"); and

WHEREAS, Parent, Merger Sub and the Company desire to make certain representations, warranties, covenants, and agreements in connection with, and establish various conditions precedent to, the Merger; and

WHEREAS, as an inducement to Parent to enter into this Agreement, certain principal stockholders of the Company are concurrently herewith entering into Voting Agreements (the "<u>Voting Agreement</u>") in substantially the form attached hereto as <u>Exhibit A</u>, whereby each such stockholder agrees to vote in favor of the Merger and all other transactions contemplated by this Agreement.

NOW, THEREFORE, in consideration of the representations, warranties, covenants and agreements set forth in this Agreement, the parties hereto, intending to be legally bound, hereby agree as follows:

Article I

The Merger

- 1.1 <u>The Merger</u>. At the Effective Time (as defined in Section 1.3 hereof), subject to the terms and conditions of this Agreement, Merger Sub shall be merged with and into the Company, the separate existence of Merger Sub shall cease, and the Company shall continue as the surviving corporation. The Company, in its capacity as the corporation surviving the Merger, is hereinafter sometimes referred to as the "<u>Surviving Corporation</u>".
- 1.2 Effect of Merger. At the Effective Time, the effect of the Merger shall be as provided in this Agreement and Section 259 of the DGCL. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time, all the properties, rights, privileges, powers and franchises of the Company and Merger Sub shall vest in the Surviving Corporation, and all debts, liabilities and duties of the Surviving Corporation.
- 1.3 Effective Time. Subject to the terms and conditions of this Agreement, the parties hereto will cause a copy of the Certificate of Merger, in substantially the form attached hereto as Exhibit B (the "Certificate of Merger") to be executed, delivered and filed with the Secretary of State of the State of Delaware in accordance with the applicable provisions of the DGCL at the time of the Closing (as defined in Section 1.7 hereof). The Merger shall become effective upon filing of the Certificate of Merger with the Secretary of State of the State of Delaware, or at such later time as may be agreed to by the parties and set forth in the Certificate of Merger. The time of effectiveness of the Merger is herein referred to as the "Effective Time". The day on which the Effective Time occurs is herein referred to as the "Effective Date".
- 1.4 Certificate of Incorporation; Bylaws. At the Effective Time, the Certificate of Incorporation of the Surviving Corporation shall be amended and restated in its entirety to be identical to the Certificate of Incorporation of Merger Sub as in effect immediately prior to the Effective Time, and such amended Certificate of Incorporation shall be the Certificate of Incorporation of the Surviving Corporation until thereafter amended in accordance with the DGCL and as provided in such Certificate of Incorporation; *provided*, *however*, that, at the Effective Time, Section 1 of the Certificate of Incorporation of the Surviving Corporation shall be amended and restated in its entirety to read as follows: "Name. The name of the corporation is LGC Wireless, Inc." and Section 4 of the Certificate of Incorporation of the Surviving Corporation shall be amended and restated in its entirety to read as follows: "Capital Stock. The total number of shares that the corporation is authorized to issue is 150,000,000, par value \$0.01 per share, all of which shares are designated as common stock." Immediately after the Effective Time, the authorized capital stock of the Surviving Corporation shall consist of 150,000,000 shares of common stock, par value \$0.01 per share. At the Effective Time, the Surviving Corporation's Bylaws shall be amended and restated in their entirety to be identical to the Bylaws of Merger Sub as in effect immediately prior to the Effective Time and such Bylaws shall be the bylaws of the Surviving Corporation until thereafter amended in accordance with the DGCL and as provided in such Bylaws.
- 1.5 <u>Directors and Officers</u>. From and after the Effective Time, the directors of the Surviving Corporation shall be the Persons who were the directors of Merger Sub immediately prior to the Effective Time, and the officers of the Surviving Corporation shall be the Persons who were the officers of Merger Sub immediately prior to the Effective Time. Said directors and officers of the Surviving Corporation shall hold office for the term specified in, and subject to the provisions contained in, the Certificate of Incorporation and Bylaws of the Surviving Corporation and applicable law. If, at or after the Effective Time, a vacancy shall exist on the board of directors or in any of the offices of the Surviving Corporation, such vacancy shall be filled in the manner provided in the Certificate of Incorporation and Bylaws of the Surviving Corporation.
- 1.6 <u>Taking of Necessary Action</u>; <u>Further Action</u>. Parent, Merger Sub and the Company, respectively, shall each use its or their best efforts to take all such action as may be necessary or appropriate to effectuate the Merger under the DGCL at the time specified in Section 1.3 hereof. If, at any time after the Effective Time, any further action is necessary or desirable to carry out the provisions of this Agreement and to vest the Surviving Corporation with full right, title and possession to all properties, rights, privileges, powers and franchises of either of the Constituent Corporations granted pursuant to the Merger, the officers of the Surviving Corporation are fully authorized in the name of each Constituent Corporation or otherwise to take, and shall take, all such lawful and necessary action.

1.7 <u>The Closing</u>. The closing of the transactions contemplated by this Agreement (the "<u>Closing</u>") will take place at the offices of Dorsey & Whitney LLP, 50 South Sixth Street, Minneapolis, Minnesota, within three business days after the date on which the last of the conditions set forth in Article VII hereof, other than the delivery of any documents required to be made at the Closing, shall have been satisfied or waived, or at such other place and on such other date as is mutually agreeable to Parent and the Company (the "<u>Closing Date</u>"). The Closing will be effective as of the Effective Time.

Article II

Conversion of Securities

- 2.1 <u>Definitions</u>. The following terms used in this Article II shall have the following meanings:
- (a) "Company Capital Stock" means, collectively, the Series A-1 Preferred Stock, the Series B-1 Preferred Stock and the Company Common Stock.
 - (b) "Company Common Stock" means the common stock, par value \$0.0001 per share, of the Company.
- (c) "Company Option" means any unexercised option to purchase Company Common Stock issued pursuant to the Company's 1997 Stock Plan, whether vested or unvested.
 - (d) "Company Warrant" means any warrant to purchase Company Common Stock, whether or not exercisable.
- (e) "<u>Escrow Percentage</u>" means the result, expressed as a percentage to three decimal points, of \$15,500,000, divided by the sum of the aggregate of the Merger Consideration payable to (i) the holders of Company Capital Stock and (ii) the holders of Qualified Warrants.
- (f) "<u>Exchange Ratio</u>" means the quotient obtained by dividing (i) the Common Price Per Share by (ii) the weighted average closing sale price for a share of common stock of Parent as quoted on the Nasdaq Global Select Market during normal trading hours for the ten consecutive trading day period ending one business day prior to the Closing Date.
 - (g) "Out-of-the-Money Option" means a Company Option set forth in Schedule 2.1(g) hereto.
 - (h) "Out-of-the-Money Warrant" means a Company Warrant set forth in Schedule 2.1(h) hereto.
- (i) "<u>Purchase Price</u>" means an amount equal (a) \$155,000,000 <u>less</u> (b) the Transaction Expenses <u>less</u> (c) the Rolling Option Incremental Value <u>plus</u> (d) the aggregate exercise prices of (1) the Qualified Options cancelled pursuant to Section 2.4(a) and converted into the right to receive the Option Spread, (2) the Rolling Options cancelled pursuant to Section 2.4(b) and converted into the right to receive Parent Rollover Options and (3) Qualified Warrants cancelled pursuant to Section 2.4(d) and converted into the right to receive the Warrant Spread.
- (j) "<u>Qualified Option</u>" means a Company Option with respect to which the holder has elected to receive the consideration described in Section 2.4(a) pursuant to a validly executed consent substantially in the form set forth in <u>Exhibit D-1</u>, which consent is in full force and effect at the Effective Time. Solely for the purposes of this Agreement, any Company Option for which a validly executed consent substantially in the form set forth in <u>Exhibit D-1</u> has not been obtained prior to the Effective Time shall be treated as if it were a Qualified Option.
 - (k) "Qualified Warrant" means a Company Warrant set forth in Schedule 2.1(k) hereto.

- (l) "<u>Rolling Option</u>" means a Company Option with respect to which the holder has elected to receive the consideration described in Section 2.4(b) pursuant to a validly executed consent substantially in the form set forth in Exhibit D-1, which consent is in full force and effect at the Effective Time.
 - (m) "Rolling Option Incremental Value" means an amount equal to \$1,000,000.
- (n) "Series A-1 Preferred Stock" means the Series A-1 mandatorily redeemable convertible preferred stock, par value \$0.0001 per share, of the Company.
- (o) "<u>Series B-1 Preferred Stock</u>" means the Series B-1 mandatorily redeemable convertible preferred stock, par value \$0.0001 per share, of the Company.
- (p) "<u>Transaction Expenses</u>" means (i) the aggregate fees and expenses payable to the Company's financial and legal advisors in connection with the transactions contemplated by this Agreement <u>plus</u> (ii) the aggregate amount, not to exceed the maximum amount set forth on *Schedule 2.1(p)* hereto, paid pursuant to the Company's Management and Employee Bonus Plan.
- 2.2 <u>Conversion of Securities</u>. By virtue of the Merger and without any action on the part of Parent, Merger Sub, the Company or any holder of shares of Company Capital Stock, the following shall occur in the following order and priority:

(a) At the Effective Time:

- (i) each share of Series B-1 Preferred Stock issued and outstanding immediately prior to the Effective Time (other than (A) any shares of Series B-1 Preferred Stock to be canceled pursuant to Section 2.2(b) hereof and (B) any Dissenting Shares) will be canceled and extinguished and be converted automatically into the right to receive \$2.34 (the "Series B-1 Price Per Share"), of which an amount equal to the product of (X) the Series B-1 Price Per Share, times (Y) the Escrow Percentage will be delivered to the Escrow Agent by Parent pursuant to Section 2.5 hereof (such amount to be referred to herein as the "Per Share Series B-1 Escrow Amount" and the difference between the Series B-1 Price Per Share and the Per Share Series B-1 Escrow Amount is to be referred to herein as the "Closing Price Per Series B-1 Share");
- (ii) if, and only to the extent that, prior to the Effective Time, the holders of the required percentage of the outstanding shares of the Series A-1 Preferred Stock have not consented to the conversion of shares of Series A-1 Preferred Stock into shares of Company Common Stock in accordance with the Company's Charter Documents, each share of Series A-1 Preferred Stock issued and outstanding immediately prior to the Effective Time (other than (A) any shares of Series A-1 Preferred Stock to be canceled pursuant to Section 2.2(b) hereof and (B) any Dissenting Shares) will be canceled and extinguished and be converted automatically into the right to receive \$1.2656 (the "Series A-1 Price Per Share"), of which an amount equal to the product of (X) the Series A-1 Price Per Share, times (Y) the Escrow Percentage will be delivered to the Escrow Agent by Parent pursuant to Section 2.5 hereof (such amount to be referred to herein as the "Per Share Series A-1 Escrow Amount" and the difference between the Series A-1 Price Per Share and the Per Share Series A-1 Escrow Amount is to be referred to herein as the "Closing Price Per Series A-1 Share"); and
- (iii) each share of Company Common Stock (including each share of Company Common Stock, if any, issued upon the conversion, prior to the Effective Time, of shares of Series A-1 Preferred Stock into shares of Company Common Stock in accordance with the Company's Charter Documents) issued and outstanding immediately prior to the Effective Time (other than (A) any shares of Company Common Stock to be canceled pursuant to Section 2.2(b) hereof and (B) any Dissenting Shares) will be canceled and extinguished and be converted automatically into the right to receive an amount (the "Common Price Per Share") equal to the result of (i) the excess, if any, of the Purchase Price over the aggregate of the amounts paid pursuant to

Sections 2.2(a)(i) and 2.2(a)(ii) above, divided by (ii) the sum of (a) the number of outstanding shares of Company Common Stock, (b) the number of shares of Company Common Stock issuable upon exercise of outstanding Company Options (other than any outstanding Out-of-the-Money Options) and (c) the number of shares of Company Common Stock issuable upon exercise of outstanding Company Warrants (other than any outstanding Out-of-the-Money Warrants), of which an amount equal to the product of (X) the Common Price Per Share, times (Y) the Escrow Percentage will be delivered to the Escrow Agent by Parent pursuant to Section 2.5 hereof (such amount to be referred to herein as the "Per Share Common Escrow Amount" and the difference between the Common Price Per Share and the Per Share Common Escrow Amount is to be referred to herein as the "Closing Price Per Common Share").

- (b) At the Effective Time, all shares of Company Capital Stock that are owned by the Company as treasury stock and each share of Company Capital Stock owned by any direct or indirect wholly owned subsidiary of the Company immediately prior to the Effective Time shall be canceled and extinguished without any conversion thereof.
- (c) At the Effective Time, each share of common stock, \$0.01 par value, of Merger Sub ("<u>Merger Sub Common Stock</u>") issued and outstanding immediately prior to the Effective Time shall be converted into and exchanged for 110,000 validly issued, fully paid and nonassessable shares of common stock, \$0.01 par value, of the Surviving Corporation, and the Surviving Corporation shall be a wholly owned subsidiary of Parent. Each stock certificate of Merger Sub evidencing ownership of any such shares shall continue to evidence ownership of such shares of capital stock of the Surviving Corporation.

2.3 Exchange of Certificates.

- (a) Promptly, and in any event no later than three business days after the Effective Time, the Parent shall cause to be mailed to each holder of record of a certificate or certificates which immediately prior to the Effective Time represented outstanding shares of Company Capital Stock (the "Certificates") (i) a letter of transmittal (which shall specify that delivery shall be effected, and risk of loss and title to the Certificates shall pass, only upon proper delivery of the Certificates to the Parent, and shall be in such form and have such other provisions as Parent may reasonably specify and which shall be reasonably acceptable to the Company) and (ii) instructions for use in effecting the surrender of the Certificates in exchange for the Closing Price Per Series B-1 Share, Closing Price Per Series A-1 Share or the Closing Price Per Common Share, as appropriate. Upon surrender of a Certificate for cancellation to Parent, together with such letter of transmittal, duly completed and validly executed, and such other documents as may be reasonably required pursuant to such instructions, the holder of such Certificate shall be entitled to receive cash in an amount equal to the number of shares of Company Capital Stock represented by such certificate multiplied by the Closing Price Per Series B-1 Share, Closing Price Per Series A-1 Share or the Closing Price Per Common Share, as appropriate (the "Aggregate Closing Price Per Certificate"), and the Certificate so surrendered shall forthwith be canceled. Until surrendered as contemplated by this Section 2.3, each Certificate that, prior to the Effective Time, represented shares of Company Capital Stock will be deemed from and after the Effective Time, to evidence the right only to receive the appropriate price per share for the shares of Company Capital Stock represented thereby multiplied by the number of such shares, without interest.
- (b) Neither Parent nor the Surviving Corporation shall be liable to any holder of shares of Company Capital Stock for any amount properly delivered to a public official in compliance with any abandoned property, escheat or similar law.
- (c) At the Effective Time, the stock transfer books of the Company shall be closed and there shall be no further registration of transfers of shares of Company Capital Stock thereafter on the records of the Company. From and after the Effective Time, the holders of certificates representing shares of Company Capital Stock outstanding immediately prior to the Effective Time shall cease to have any rights with respect to such shares of Company Capital Stock, except as otherwise specifically provided in this Agreement or by law.

- (d) If any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact, in form and substance reasonably acceptable to Parent, by the Person claiming such Certificate to be lost, stolen or destroyed, and complying with such other conditions as Parent may reasonably impose (including the execution of an indemnification undertaking in favor of Parent with respect to the Certificate alleged to be lost, stolen or destroyed), Parent will deliver to such Person, the Aggregate Closing Price Per Certificate in respect of such lost, stolen or destroyed Certificate.
- (e) If any holder of Dissenting Shares becomes entitled to receive payment for such shares pursuant to applicable law and Section 2.9 hereof, such payment will be made by the Surviving Corporation in accordance with Section 2.9 hereof.

2.4 Company Options/Company Warrants.

- (a) Each unexercised Qualified Option outstanding immediately prior to the Effective Time for which a validly executed consent substantially in the form set forth in Exhibit D-1 hereto has been received prior to the Effective Time will be cancelled and extinguished and be converted into and become a right to receive an amount equal to (A) the Common Price Per Share in excess of the exercise price of such Qualified Option (the "Option Spread") times (B) the number of shares of Company Common Stock for which such Qualified Option is exercisable. Each unexercised Qualified Option outstanding immediately prior to the Effective Time for which no validly executed consent has been received prior to the Effective Time shall remain a Qualified Option, exercisable in accordance with its terms, until such time as the holder thereof shall validly execute and deliver to Parent or to Surviving Corporation a consent, in form reasonably acceptable to Parent, electing to receive the consideration described in this Section 2.4(a), at which point, such Qualified Option shall be cancelled and extinguished and become a right to receive an amount equal to the Option Spread times the number of shares of Company Common Stock for which such Qualified Option is exercisable.
- (b) Each unexercised Rolling Option outstanding immediately prior to the Effective Time will be cancelled and extinguished and there will be substituted in exchange therefor an option ("Parent Rollover Option") to purchase such number of shares of common stock of Parent equal to the number of shares of Company Common Stock that were issuable upon exercise of each such Rolling Option immediately prior to the Effective Time multiplied by the Exchange Ratio (rounded down to the nearest whole number of shares of the common stock of Parent), and the per share exercise price of the shares of common stock of the Parent issuable upon the exercise of such Parent Rollover Option will be equal to the exercise price per share of the Company Common Stock at which such Rolling Option was exercisable immediately prior to the Effective Time divided by the Exchange Ratio (rounded up to the nearest whole cent), with such substitution to be carried out in accordance with the requirements of Treasury Regulations, Section 1.424-1(a) and Section 409A of the Code (and all parties agree to report the substitution accordingly).
- (c) Each unexercised Out-of-the-Money Option outstanding immediately prior to the Effective Time will be cancelled and extinguished without the right to receive any cash or other payment.
- (d) Each unexercised Qualified Warrant outstanding immediately prior to the Effective Time will be canceled and extinguished and be converted into and become a right to receive an amount equal to (A) the Common Price Per Share in excess of the exercise price of such Qualified Warrant (the "Warrant Spread") times (B) the number of shares of Company Common Stock for which such Qualified Warrant is exercisable, provided that, an amount equal to the product of (X) the Warrant Spread, times (Y) the Escrow Percentage (such amount to be referred to herein as the "Per Warrant Escrow Amount") times (Z) the number of shares of Company Common Stock for which such Qualified Warrant is exercisable will be delivered to the Escrow Agent by Parent pursuant to Section 2.5 hereof. The difference between the Warrant Spread and the Per Warrant Escrow Amount is referred to herein as the "Closing Price Per Warrant".

- (e) Each unexercised Out-of-the-Money Warrant outstanding immediately prior to the Effective Time will be cancelled and extinguished without the right to receive any cash or other payment.
- (f) As soon as practicable after the Effective Time, provided that prior to the Effective Time Parent has received a validly executed consent in the form attached hereto as Exhibit D-1 containing an election to receive the consideration described in Section 2.4(b) with respect to such Rolling Option, Parent will substitute a Parent Rollover Option for each Rolling Option, with such substitution to be carried out in compliance with the requirements of Treasury Regulations, Section 1.424-1(a) and Section 409A of the Code (and all parties agree to report the substitution accordingly). As soon as practicable after the Effective Time, but subject to any withholding tax requirement, the Surviving Corporation will pay to the holder of each Qualified Option an amount equal to (A) the Option Spread times (B) the number of shares of Company Common Stock for which each such Qualified Option was exercisable, without interest, provided, however, that no holder of a Company Option shall be entitled to receive the amount described in this sentence unless and until such holder has validly executed and delivered to Parent a consent substantially in the form attached hereto as Exhibit D-1. As soon as practicable after the Effective Time, but subject to any withholding tax requirement and provided that the amendment described in Section 7.2(g) has been validly completed, the Surviving Corporation will pay to the holder of each Qualified Warrant an amount equal to (X) Closing Price Per Warrant times (Y) the number of shares of Company Common Stock for which such Qualified Warrant was exercisable, without interest. All other Company Options and Company Warrants, including all Out-of-the-Money Options and Out-of-the-Money Warrants, will be canceled as of the Effective Time without the right to receive any cash or other payment.
- (g) Prior to the Effective Time, the Company will take all action that may be necessary (under the plans and agreements pursuant to which Company Options are outstanding, the agreements pursuant to which Company Warrants are outstanding and otherwise) to effectuate the provisions of this Section 2.4 and to ensure that, from and after the Closing Date, holders of Company Options and/or Company Warrants have no rights with respect thereto other than those specifically provided in this Section 2.4.
- 2.5 Escrow. An amount equal to \$15.500.000 (the "Escrow Amount") shall be deposited by Parent with U.S. Bank National Association, as escrow agent selected by Parent (the "Escrow Agent"), to be held in escrow (the "<u>Escrow Fund</u>") in an account (the "<u>Escrow Account</u>") pursuant to the terms of the Escrow Agreement (the "<u>Escrow</u> Agreement") among Parent, the Stockholders' Representative and the Escrow Agent in substantially the form attached hereto as Exhibit C. Distributions of any amounts from the Escrow Account shall be governed by the terms and conditions of the Escrow Agreement. Specific reference is made to (i) Section 5 of the Escrow Agreement, which shall govern the disbursement of funds from the Escrow Account to Parent for satisfaction of indemnification claims under Article IX hereof, (ii) Sections 6(a)-(b) of the Escrow Agreement, which shall govern the disbursement of funds from the Escrow Account to the holders of Company Capital Stock and Qualified Warrants following the 15-month anniversary of the Closing Date and (iii) Sections 6(c)-(d) of the Escrow Agreement which shall govern the disbursement of funds from the Escrow Account to the holders of Company Capital Stock and Qualified Warrants following the three-year anniversary of the Closing Date. The portion of the Escrow Amount contributed on behalf of each holder of Company Capital Stock or Qualified Warrants shall be the same proportion of the Escrow Amount equal to the aggregate amount of cash such holder would otherwise be entitled to receive under Section 2.2 hereof by virtue of such holder's ownership of Company Capital Stock or Qualified Warrants to the total Merger Consideration received by all such holders. For the avoidance of doubt, Qualified Options and Rolling Options (and the respective Parent Rollover Options) shall not be subject to the Escrow Fund.
- 2.6 <u>Adjustment of Shares</u>. If, during the period between the date of this Agreement and the Effective Time, any change in the number, classes or series of outstanding shares of Company Capital Stock shall occur, including by reason of any reclassification, recapitalization, stock dividend, stock split or combination, exchange or readjustment of such shares of Company Capital Stock, or any stock dividend thereon with a record date during such period, the Series B-1 Price Per Share, the Series A-1 Price Per Share, the Common Price Per Share, the Option Spread, the Warrant Spread and the respective Per Share Escrow Amounts or Per Warrant Escrow Amounts for each of the

foregoing, and any other amounts payable pursuant to this Agreement, as the case may be, shall be appropriately adjusted.

- 2.7 <u>Merger Consideration</u>. Notwithstanding anything to the contrary contained in this Agreement, the aggregate value delivered by Parent pursuant to this Agreement (the "<u>Merger Consideration</u>") shall not exceed \$155,000,000 <u>less</u> the Transaction Expenses.
- 2.8 Tax Withholding. Each of the Surviving Corporation, Parent and Escrow Agent shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement to any holder of Company Capital Stock, Company Option or Company Warrant such amounts as it is required to deduct and withhold with respect to the making of such payment under the Internal Revenue Code of 1986, as amended (the "Code"), or any provision of state, local or foreign Tax law. To the extent that amounts are so withheld by the Surviving Corporation, Parent or Escrow Agent, as the case may be, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to such holder in respect of which such deduction and withholding was made by the Surviving Corporation, Parent or Escrow Agent, as the case may be. To the extent any such amount is withheld, the Surviving Corporation, Parent or Escrow Agent, as applicable, shall pay such withheld amount to the appropriate Governmental Entity/ Entities on a timely basis and shall indemnify and hold harmless any holder of Company Capital Stock, Company Options or Company Warrants for the payment of any such withheld amount.

2.9 Dissenting Shares.

- (a) Notwithstanding anything in this Agreement to the contrary, if Section 262 of the DGCL is applicable to the Merger, shares of Series B-1 Preferred Stock, Series A-1 Preferred Stock and Company Common Stock that are issued and outstanding immediately prior to the Closing Date and which are held by stockholders who have not executed a written consent approving the Merger, who will have delivered a written demand for the appraisal of such shares in the manner provided in Section 262 of the DGCL and who, as of the Closing Date, will not have effectively withdrawn or lost such right to dissenters' rights ("Dissenting Shares") shall not, upon the Closing, be converted into or represent a right to receive the Merger Consideration pursuant to Section 2.2 hereof, but the holders thereof shall be entitled only to such rights as are granted by Section 262 of the DGCL. Each holder of Dissenting Shares who becomes entitled to payment for such shares pursuant to Section 262 of the DGCL shall receive payment therefor from the Surviving Corporation in accordance with the DGCL.
- (b) The Company will give Parent (i) prompt notice of any written demand for appraisal, any withdrawal of a demand for appraisal and any other instrument served pursuant to Section 262 of the DGCL received by the Company and (ii) the opportunity to participate, in consultation with the Company, in all negotiations and proceedings with respect to demands for appraisal under such Section 262 of the DGCL. The Company will not, except with the prior written consent of Parent, voluntarily make any payment with respect to any demand for appraisal or offer to settle or settle any such demand.

2.10 Post-Closing Adjustment to Purchase Price.

(a) The "<u>Measurement Date</u>" shall be the later to occur of (i) November 30, 2007 and (ii) the last business day of the month in which the Closing Date occurs. Within 60 days after the Measurement Date, Parent will prepare and deliver to Stockholders' Representative a consolidated balance sheet (the "<u>Measurement Date Balance Sheet</u>") for the Surviving Corporation and the Subsidiaries as of the close of business on the Measurement Date (determined on a stand alone basis without giving effect to the transactions contemplated by this Agreement, in accordance with GAAP applied on a basis consistent with the preparation of the Latest Financial Statements). The Measurement Date Balance Sheet will include a determination of the Measurement Date Net Assets of the Surviving Corporation and the Subsidiaries as of the close of business on the Measurement Date. "<u>Measurement Date Net Assets</u>" means the excess of total assets over total liabilities shown on the Measurement Date Balance Sheet. Parent will make the workpapers and back-up materials used in preparing the Measurement Date Balance Sheet available to Stockholders'

Representative and Stockholders' Representative's accountants and other representatives at reasonable times and upon reasonable notice during (i) the review by Stockholders' Representative of the Measurement Date Balance Sheet and (ii) the resolution by Parent and Stockholders' Representative of any objections to the Measurement Date Balance Sheet.

- (b) Stockholders' Representative may object to the Measurement Date Balance Sheet on the basis that it was not prepared in accordance with GAAP applied on a basis consistent with the preparation of the Latest Financial Statements or that the calculation of Measurement Date Net Assets contains mathematical errors. If Stockholders' Representative has any objections to the Measurement Date Balance Sheet or the Measurement Date Net Assets, Stockholders' Representative will deliver a detailed statement describing such objections (the "Statement of Objection") to Parent within 20 days after receiving the Measurement Date Balance Sheet. Parent and Stockholders' Representative will attempt in good faith to resolve any such objections. If Parent and Stockholders' Representative do not reach a resolution of all objections within 30 days after Parent has received the Statement of Objection, Parent and Stockholders' Representative will select a mutually acceptable independent accounting firm (the "Independent Accountant") to resolve any remaining objections. If Parent and Stockholders' Representative are unable to agree on the choice of an accounting firm, they will select a nationally recognized accounting firm by lot (after excluding the regular outside accounting firms of Parent and the Company). The selected Independent Accountant will resolve any such objections and determine, in accordance with GAAP applied on a basis consistent with the preparation of the Latest Financial Statements, the amounts to be included in the Measurement Date Balance Sheet and the Measurement Date Net Assets. The parties will provide the Independent Accountant, within 10 days of its selection, with a definitive statement of the position of each party with respect to each unresolved objection and will advise the Independent Accountant that the parties accept the Independent Accountant as the appropriate Person to interpret this Agreement for all purposes relevant to the resolution of the unresolved objections. Parent will provide the Independent Accountant access to the books, records and personnel of each of the Company and the Subsidiaries. The Independent Accountant will have 30 days to carry out a review of the unresolved objections and prepare a written statement of its determination regarding each unresolved objection. The determination of any Independent Accountant so selected will be set forth in writing and will be conclusive and binding upon the parties. Parent will revise the Measurement Date Balance Sheet and the determination of the Measurement Date Net Assets as appropriate to reflect the resolution of any objections to the Measurement Date Balance Sheet pursuant to this Section 2.10.
- (c) If Parent and Stockholders' Representative submit any unresolved objections to the Independent Accountant for resolution as provided in this Section 2.10, Parent and Stockholders' Representative will each bear their respective costs and expenses and will share equally in the fees and expenses of the Independent Accountant.
- (d) If the Measurement Date Net Assets as finally determined is less than the Latest Balance Sheet Net Assets Amount, then, within 10 business days after the date on which the Measurement Date Net Assets is finally determined pursuant to this Section 2.10, an amount equal to the difference between (x) the Measurement Date Net Assets as finally determined and (y) Latest Balance Sheet Net Assets Amount will be withdrawn by the Escrow Agent from the Escrow Account and paid to Parent. "Latest Balance Sheet Net Assets Amount" means \$11,157,000.
- (e) Any payment made pursuant to this Section 2.10 will not preclude any remedy provided in this Agreement or otherwise for any breach of representation, warranty or agreement, and the remedy provided in this Agreement for any breach of representation, warranty or agreement or otherwise will not preclude the adjustment provided in this Section 2.10.
- (f) Judgment upon the award rendered by the accounting firm may be entered in any court of competent jurisdiction.
- (g) Parent agrees that, during the period commencing on the Closing Date and ending on the Measurement Date, it will not take any action specifically designed to impair the Measurement Date Net Assets and has the primary purpose of causing an adjustment pursuant to Section 2.10(d) hereof. The Company, however, acknowledges that the

transactions contemplated in this Agreement constitute a transfer of control to Parent and that, following the Closing Date, Parent and its representatives will exercise their business judgment regarding the operations of the Surviving Corporation in a manner consistent with Parent's duties to its shareholders.

Article III

Representations and Warranties of the Company

The Company hereby represents and warrants to Parent, and acknowledges that Parent is relying upon the following representations and warranties, that, except as set forth in the Disclosure Schedule:

- 3.1 Incorporation; Corporate Power and Authority.
- (a) Each of the Company and the Subsidiaries is a corporation duly incorporated (or the equivalent for those Subsidiaries in foreign jurisdictions), validly existing, duly registered (in each case, if applicable) and in good standing under the laws of the jurisdiction of its organization and has all requisite corporate power and authority necessary to carry on its business as now being conducted and to own, lease and operate its assets. The Company and each of the Subsidiaries is duly qualified as a foreign corporation to do business in every jurisdiction in which the nature of its business or its ownership of property requires it to be so qualified, except for those jurisdictions in which the failure to be so qualified would not, individually or in the aggregate, have a Material Adverse Effect (as defined below). Section 3.1 of the Disclosure Schedule sets forth a true and complete list, by corporation, of all jurisdictions in which the Company and each of the Subsidiaries is qualified and in good standing, if applicable. As used herein, the term "Material Adverse Effect" means any change, effect, event or condition that (i) has had or could, individually or in the aggregate, reasonably be expected to have a material adverse effect on the business, assets (including intangible assets), results of operations, or condition (financial or otherwise), of the Company and its Subsidiaries, taken as a whole, or on the ability of the Company and its Subsidiaries, taken as a whole, to materially achieve the Company's revenue projections, recorded on a GAAP basis, pertaining to the calendar year ending December 31, 2007, as delivered to Parent on July 19, 2007, other than any such change, effect, event or condition that results or arises from (A) changes or conditions affecting the industry in which the Company markets its products and services generally, except to the extent such changes or conditions materially disproportionately affect the Company and the Subsidiaries. taken as a whole, (B) changes in general economic, regulatory or political conditions, except to the extent such changes or conditions materially disproportionately affect (relative to other participants in the industry in which the Company markets its products and services) the Company and the Subsidiaries, taken as a whole, (C) changes in GAAP, (D) any change, effect, event or condition exclusively relating to any acts of terrorism, sabotage, military action or war or (E) any change, effect, event or condition arising solely and directly out of the execution, delivery, performance or disclosure of this Agreement or the transactions contemplated hereby, including any impact thereof on relationships, contractual or otherwise, with customers, suppliers or distributors, and that would not have resulted or arisen in the absence of such execution, delivery, performance (in accordance with its terms) or disclosure (excluding from this subsection (E) all changes, effects, events or conditions that existed prior to the date hereof), or (ii) would prevent or materially delay the Company's ability to consummate the Merger or the other transactions contemplated hereby.
- (b) Neither the Company nor any Subsidiary is in violation of any of the provisions of its Certificate of Incorporation or other applicable charter document (any such document hereinafter referred to as its "*Charter Documents*") or Bylaws or other applicable governing document (any such documents hereinafter referred to as its "*Governing Documents*"). The Company has made available to Parent in the electronic data room hosted by DLA Piper and known as "Project Red Wings" (the "*Data Room*"), accurate and complete copies of the respective Charter Documents and Governing Documents, as currently in effect, of each of the Company and the Subsidiaries.
- 3.2 <u>Subsidiaries</u>. The Company is the record and beneficial owner of the outstanding shares of capital stock of each of the entities listed (and in the amount and ownership percentage shown) in Section 3.2(a) of the Disclosure

Schedule (each such entity, a "Subsidiary" and together, the "Subsidiaries"). The Company maintains the branch offices and representative offices listed in Section 3.2(b) of the Disclosure Schedule. Each Subsidiary, branch office and representative office has complied in all material respects with the laws of each jurisdiction in which such Subsidiary, branch office or representative office operates. Neither the Company nor any of its Subsidiaries owns, controls or holds with the power to vote, directly or indirectly, of record, beneficially or otherwise, any capital stock or any equity or ownership interests in any corporation, partnership, association, joint venture or other entity, except for the Subsidiaries. All of the outstanding shares of capital stock and all equity or other interests of each of the Company's Subsidiaries have been duly authorized and are validly issued, fully paid and nonassessable and were not issued in violation of any preemptive rights or comparable rights. There are no proxies or similar obligations with respect to any interest in each Subsidiary and all interests in each Subsidiary are owned by the Company or another Subsidiary of the Company in each instance free and clear of all Liens with respect thereto.

3.3 Capitalization.

- (a) The authorized capital stock of the Company consists of 190,792,791 shares of capital stock, of which 143,000,000 shares are designated as Company Common Stock and 47,792,791 shares are designated preferred stock, par value \$0.0001 per share. 31,126,125 shares of Series A-1 Preferred Stock are issued and outstanding; 16,480,768 shares of Series B-1 Preferred Stock are issued and outstanding; and 31,777,493 shares of Company Common Stock are issued and outstanding. The Company has no other securities outstanding that are or would be entitled to vote on the transactions contemplated by this Agreement. The issued and outstanding shares of Company Capital Stock are duly authorized, validly issued, fully paid and nonassessable, and are free of preemptive rights or any other third party rights. All issued and outstanding shares of Company Capital Stock have been offered, sold and delivered by the Company in compliance with all federal and state securities and state blue sky laws. No shares of the Company Capital Stock have been issued in violation of any preemptive rights, rights of first refusal or similar rights. The rights and privileges of each class of Company Capital Stock are set forth in the Company's Certificate of Incorporation.
- (b) As of the date of this Agreement, Company Options with respect to 13,514,885 shares of Company Common Stock are outstanding. Section 3.3(b) of the Disclosure Schedule lists, as of the date hereof, (i) the name of each holder of an outstanding Company Option, (ii) such holder's address per the Company's records, which, to the Company's knowledge of such matter, are complete and correct, (iii) whether such holder is an employee of the Company and, with respect to each Company Option held, (iv) the date of grant of such Company Option, (v) the number of shares of Company Common Stock subject to such Company Option, (vi) the exercise price of such Company Option, (vii) the vesting schedule (and any provisions for acceleration or deferral of vesting) for such Company Option, (viii) the extent vested as of the date of this Agreement and (ix) whether such Company Option is an "incentive stock option." All issued and outstanding Company Options have been offered, sold and delivered by the Company in compliance with all federal and state securities and state blue sky laws.
- (c) As of the date of this Agreement, Company Warrants with respect to 15,673,551 shares of Company Common Stock are outstanding. Section 3.3(c) of the Disclosure Schedule lists, as of the date hereof, (i) the name of each holder of an outstanding Company Warrant, (ii) such holder's address per the Company's records, which, to the Company's knowledge of such matter, are complete and correct, (iii) the number of shares of Company Common Stock subject to such Company Warrant, (iv) the exercise price of such Company Warrant and (v) the expiration date of such Company Warrant. All issued and outstanding Company Warrants have been offered, sold and delivered by the Company in compliance with all federal and state securities and state blue sky laws.
- (d) All shares of Company Common Stock issuable upon exercise of the Company Options and Company Warrants have been offered in compliance with all federal and state securities and state blue sky laws and, upon issuance in accordance with their terms, will be duly authorized, validly issued, fully paid and nonassessable and will be free of preemptive rights or any other third party rights.

- (e) Each Company Option (i) has been granted in accordance with the terms of the applicable Company stock option plan and the relevant stock option agreement, (ii) has been granted with an exercise price at least equal to the fair market value of the Company Common Stock on the grant date, and (iii) has a grant date that is the date the option would be considered granted for tax, corporate law and under generally accepted accounting principles (that is, no Company Option has been "backdated").
- (f) Except for the Company Options and Company Warrants listed in Sections 3.3(b) and 3.3(c) of the Disclosure Schedule, the Series A-1 Preferred Stock and the Series B-1 Preferred Stock and the Convertible Notes, there is no option, warrant, call, subscription, convertible security, right (including preemptive right) or Contract of any character to which the Company is a party or by which it is bound obligating the Company to issue, exchange, transfer, sell, repurchase, redeem or otherwise acquire any capital stock of the Company or obligating the Company to grant, extend, accelerate the vesting of or enter into any such option, warrant, call, subscription, convertible security, right or Contract. There are no outstanding or authorized stock appreciation, phantom stock or similar rights with respect to the Company. Except as contemplated by this Agreement, the Company is not a party to, and, to the Company's knowledge, no stockholder of the Company is party to, any registration rights agreements, voting trust, proxy or other Contract pertaining to the capital stock of the Company and no restrictions on transfer with respect to any capital stock of the Company exist. The share registers and the transfer of shares of the Company, copies of which have been made available to Parent in the Data Room prior to the date hereof, are up-to-date, complete and correct.
- 3.4 Execution, Delivery; Valid and Binding Agreement. The Company has all requisite corporate power and authority to execute and deliver, and perform its obligations under, this Agreement and to consummate the transactions contemplated hereby. The execution, delivery and performance of this Agreement by the Company and the consummation of the transactions contemplated hereby have been duly and validly authorized by all requisite corporate action and, other than the approval and adoption of this Agreement and the Merger by the holders of Company Capital Stock in accordance with the Company's Charter Documents and the DGCL as set forth in Section 3.26 (the "Required Stockholder Vote"), no other corporate proceedings on the Company's part are necessary to authorize the execution, delivery or performance of this Agreement. This Agreement has been duly and validly executed and delivered by the Company and constitutes the valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, and the other documents contemplated hereby, when executed and delivered by the Company, will constitute the valid and binding obligations of the Company, enforceable against the Company in accordance with their respective terms, in each case except to the extent that their enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other laws affecting the enforcement of creditors' rights generally and by general equitable principles.
- 3.5 No Violations, etc. The execution, delivery and performance of this Agreement by the Company does not and the consummation of the transactions contemplated hereby will not: (a) contravene any provision of the Certificate of Incorporation or Bylaws of the Company; (b) violate or conflict in any material respect with any federal, state, local or foreign law or any decree, writ, injunction, judgment or order of any court or administrative or other governmental body or of any arbitration award which is either applicable to, binding upon or enforceable against the Company or any of the Subsidiaries, or the business or any assets of the Company or any of the Subsidiaries; (c) conflict in any material respect with or result in any material breach of any of the provisions of, or constitute a material default (or any event which would, with the passage of time or the giving of notice or both, constitute a material default) under, result in a material violation of, result in the creation of a right of termination, amendment, modification, abandonment or acceleration under any indenture, hypothecation, mortgage, lease, license, loan agreement or other material agreement or instrument which is either binding upon or enforceable against the Company or any of the Subsidiaries; (d) result in the creation of any charge, claim, easement, covenant, equitable interest, option, lien, pledge, security interest, encumbrance, right of first refusal, restrictions of any kind, including on voting, transfer, receipt of income or exercise of any other attribute of ownership (each, without reference to materiality, a "Lien" and together, the "Liens") that is material upon the Company or any of the Subsidiaries or any of the assets of the Company or any of the Subsidiaries; or (e) require any authorization, consent, approval, exemption or other action by or notice to any federal, state, local, foreign, international or multinational entity or authority exercising executive, legislative, judicial,

regulatory, administrative or taxing functions of or pertaining to government (each, a "<u>Governmental Entity</u>") or any other third party, other than (i) in connection with the applicable requirements of the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations thereunder (the "<u>HSR Act</u>"), and any other comparable foreign merger or competition laws listed in Section 3.5 of the Disclosure Schedule, (ii) such consents, waivers, approvals, orders, authorizations, registrations, declarations and filings as may be required under applicable federal or state securities laws, (iii) consents set forth in Section 3.5 of the Disclosure Schedule.

3.6 Financial Statements.

- (a) The Company has delivered to Parent true and complete copies of (i) the audited consolidated balance sheets, as of December 31, 2005 and December 31, 2006, of the Company and the audited statements of operations, mandatory redeemable convertible preferred stock and stockholders' equity and cash flows of the Company for the years ended December 31, 2005 and December 31, 2006 (collectively, the "<u>Annual Financial Statements</u>"), and (ii) the unaudited consolidated balance sheet, as of June 30, 2007, of the Company (the "<u>Latest Balance Sheet</u>") and the unaudited consolidated statements of operations of the Company for the six-month period ended June 30, 2007 (such unaudited statements and the Latest Balance Sheet being herein referred to as the "<u>Latest Financial Statements</u>").
- (b) The Annual Financial Statements and the Latest Financial Statements are based upon the information contained in the books and records of the Company and the Subsidiaries and fairly present in all material respects the financial condition of the Company and the Subsidiaries as of the dates thereof and results of operations for the periods referred to therein. The Annual Financial Statements have been prepared in accordance with United States generally accepted accounting principles ("*GAAP*"). The Latest Financial Statements have been prepared on a basis consistent with the Annual Financial Statements and in accordance with GAAP applicable to unaudited interim financial statements (and thus may not contain all notes and may not contain prior period comparative data which are required in order for such financial statements to be prepared in accordance with GAAP), and reflect all adjustments necessary to a fair statement of the results for the interim period(s) presented (except for normally recurring year-end adjustments).
- (c) Section 3.6(c) of the Disclosure Schedule lists, and the Company has made available to Parent in the Data Room copies of the documentation creating or governing, all securitization transactions and "off-balance sheet arrangements" (as defined in Item 303(a) of Regulation S-K adopted by the Securities and Exchange Commission (the "SEC")) effected by the Company or the Subsidiaries since January 1, 2006.
- (d) Since January 1, 2006, no fraud, whether or not material, that involves management or other employees who have a significant role in the preparation of financial reports of the Company and the Subsidiaries, as a whole, has been disclosed to the Company's auditors, Board of Directors or executive management.
- 3.7 <u>Absence of Undisclosed Liabilities</u>. Neither the Company nor any Subsidiary has any liabilities (whether accrued, absolute, contingent, unliquidated, asserted, unasserted or otherwise, and whether due or to become due, and whether known or unknown), arising out of transactions or events heretofore entered into, or any action or inaction, or any state of facts existing, with respect to or based upon transactions or events heretofore occurring, except (i) as reflected in the Latest Balance Sheet, (ii) which has arisen after the date of the Latest Balance Sheet (the "*Balance Sheet Date*") in the ordinary course of business (none of which is a material uninsured liability for breach of contract, breach of warranty, tort, infringement, claim or lawsuit) or (iii) for any liability disclosed as such by any section of the Disclosure Schedule.
- 3.8 <u>Absence of Certain Developments</u>. Since the Balance Sheet Date, there has been no change in the Company or any Subsidiary which change has had or, with the passage of time, is reasonably likely to have, a Material Adverse Effect and, except for actions specifically described in or contemplated by this Agreement, including the Exhibits

thereto and any documents delivered at Closing pursuant to this Agreement, neither the Company nor any Subsidiary has:

- (a) borrowed any amount (including advances on existing credit facilities) or incurred or become subject to any liability in excess of \$500,000 individually, or \$1,000,000 in the aggregate, *except* (i) current liabilities incurred in the ordinary course of business and (ii) liabilities under contracts entered into in the ordinary course of business;
- (b) hypothecated, mortgaged, pledged or subjected to any Lien, any of its assets with a fair market value in excess of \$500,000 individually, or \$1,000,000 in the aggregate, except (i) Liens for current property taxes not yet due and payable, (ii) Liens imposed by law and incurred in the ordinary course of business for obligations not yet due to carriers, warehousemen, laborers, materialmen and the like, (iii) Liens in respect of pledges or deposits under workers' compensation laws or (iv) Liens set forth in Section 3.8 of the Disclosure Schedule (collectively, the "Permitted Liens");
- (c) sold, assigned or transferred (including transfers to any employees, affiliates or stockholders) any tangible assets of its business with a value in excess of \$100,000, except sales of inventory in the ordinary course of business, or canceled any debts or claims, except in the ordinary course of business;
- (d) sold, assigned, transferred or granted (including transfers to any employees, affiliates or stockholders) any licenses, patents, trademarks, trade names, domain names, copyrights, trade secrets or other intangible assets, other than licenses granted on a non-exclusive basis in conjunction with the sale of product in the ordinary course of business;
- (e) disclosed, to any Person other than Parent and authorized representatives of Parent, any proprietary confidential information, other than pursuant to a confidentiality agreement limiting the use or further disclosure of such information, which agreement is in full force and effect on the date hereof;
- (f) waived any rights of material value or suffered any extraordinary losses or material adverse changes in collection loss experience, whether or not in the ordinary course of business or consistent with past practice;
- (g) issued, sold or transferred any of its equity securities, securities convertible into or exchangeable for its equity securities or warrants, options or other rights to acquire its equity securities, or any bonds or debt securities other than pursuant to instruments listed in Section 3.3 of the Disclosure Schedule;
- (h) taken any other action or entered into any other transaction other than in the ordinary course of business, or entered into any transaction with any Insider (as defined in Section 3.21 hereof) other than employment arrangements otherwise disclosed in this Agreement and the Disclosure Schedule, or the transactions contemplated by this Agreement;
- (i) suffered any material theft, damage, destruction or loss of or to any property or properties owned or used by it, whether or not covered by insurance;
- (j) entered into or modified any employment, severance or similar agreements or arrangements with, or granted any bonuses, salary or benefits increases, severance or termination pay to, any employee other than in the ordinary course of business and consistent with past practice, or to any officer or consultant;
- (k) adopted or amended any bonus, profit sharing, compensation, stock option, pension, retirement, deferred compensation, employment or other employee benefit plan, trust, fund or group arrangement for the benefit or welfare of any employees, officer, director or affiliate;
- (l) made any capital expenditure or commitment therefor in excess of \$100,000 individually, or \$500,000 in the aggregate;

- (m) made any loans or advances to, or guarantees for the benefit of, any Affiliates;
- (n) made any material loans or advances to, or guarantees for the benefit of, any Persons, other than in the ordinary course of business, consistent with past practice;
- (o) acquired (by merger, exchange, consolidation, acquisition of stock or assets or otherwise) any corporation, partnership, limited liability company, joint venture or other business organization or division or material assets thereof;
- (p) made any charitable pledges the amount remaining to be paid of which individually or in the aggregate exceeds \$100,000; or
- (q) made any change in any Tax or financial accounting methods, principles, practices, periods or elections from those utilized in the preparation of the most recently filed Tax Returns (as defined in Section 3.12(l)) or in the Annual Financial Statements, except as required by GAAP, the statutory accounting principles and practices prescribed or permitted by the domiciliary state of the relevant Person, or applicable Tax law.

3.9 Title to Properties.

- (a) Neither the Company nor any Subsidiary owns any real property. The real property covered by the leases (the "<u>Leases</u>") described in Section 3.9 of the Disclosure Schedule constitutes all of the real property rented, used or occupied by the Company and the Subsidiaries (the "<u>Real Property</u>"). The Real Property has access, sufficient for the conduct of the Company's and the Subsidiaries' business as now conducted, to public roads and to all necessary utilities
- (b) The Leases are in full force and effect and the Company and each Subsidiary, as applicable, holds a valid and existing leasehold interest under each of the respective Leases. The Company has made available to Parent in the Data Room complete and accurate copies of each of its Leases, and none of the Leases has been modified in any respect, except to the extent that such modifications are disclosed by the copies made available to Parent in the Data Room. Neither the Company nor any Subsidiary is in default under any of its Leases and no circumstances exist which, if unremedied, would, either with or without notice or the passage of time or both, result in such default under any of the Leases; nor, to the knowledge of the Company, is any other party to any of the Leases in default thereunder.
- (c) The Company or one of the Subsidiaries owns good and marketable title to each of the tangible properties and tangible assets reflected on the Latest Balance Sheet or acquired since the date thereof, free and clear of all Liens, except for (i) Liens for current Taxes not yet due and payable, (ii) the Real Property subject to the Leases, (iii) personal property used by the Company and subject to lease, and (iv) assets disposed of since the Balance Sheet Date.
- (d) All of the buildings, machinery, equipment and other tangible assets that are necessary for the conduct of the Company's and the Subsidiaries' business are in good condition and repair, ordinary wear and tear excepted with respect to all of such assets, and are usable in the ordinary course of business. The Company and the Subsidiaries own, or lease under valid leases, all buildings, machinery, equipment and other tangible assets necessary for the conduct of their business as currently conducted.
- (e) Neither the Company nor any Subsidiary has received any notice of any material violation of any material applicable zoning ordinance or other law, regulation or requirement relating to the operation of any properties used in the operation of its business.
- (f) Neither the Company nor any Subsidiary has knowledge of improvements made or contemplated to be made by any public or private authority, the costs of which are to be assessed as special Taxes or charges against any of the Real Property, and there are no present assessments.

- 3.10 <u>Accounts Receivable</u>. The accounts receivable reflected on the Latest Balance Sheet are valid receivables, have arisen from bona fide transactions in the ordinary course of business, are not subject to valid counterclaims or setoffs, and, to the Company's knowledge, are collectible in accordance with their respective terms.
- 3.11 <u>Inventory</u>. The Company's and each Subsidiary's inventory of raw materials, work in process and finished products relating to its business, subject to any applicable reserves reflected on the Latest Balance Sheet and that would normally occur at the end of the applicable quarter consistent with past practices, consists of items in all material respects of a quality and quantity usable and, with respect to finished products only, salable in the ordinary course of its business.

3.12 Tax Matters.

- (a) Each of the Company and each of the Subsidiaries and any affiliated, combined or unitary group for federal or state income tax or franchise Tax purposes of which any of the Company or the Subsidiaries is a member (each of such Subsidiaries or group, a "*Tax Affiliate*" and, collectively, the "*Tax Affiliates*"), has: (i) timely filed (or has had timely filed on its behalf) all material Tax Returns required to be filed or sent by it in respect of any Taxes due or payable on or prior to the date hereof or required to be filed or sent by it by any taxing authority having jurisdiction, which Tax Returns are true and correct in all material respects and have been completed in material compliance with applicable law; (ii) timely and properly paid (or has had paid on its behalf) all material Taxes payable in respect of the Company and its Tax Affiliates, whether or not shown on such Tax Returns; and (iii) materially complied with all applicable laws relating to the withholding of Taxes and the payment thereof (including withholding of Taxes under Sections 1441 and 1442 of the Code, or similar provisions under any foreign laws) to the proper taxing authority.
- (b) There are no Liens for Taxes upon any assets of the Company or any Tax Affiliate, except Liens for Taxes not yet due or Taxes being contested in good faith.
- (c) No deficiency for any material Taxes has been proposed, asserted or assessed against the Company or any of the Tax Affiliates, in each case in writing sent to the Company or a Tax Affiliate or in an oral statement made by the proper taxing authorities to the Company or a Tax Affiliates, that has not been resolved and paid in full. No waiver, extension or comparable consent given by the Company or the Tax Affiliates regarding the application of the statute of limitations or the period for assessment or reassessment with respect to any Taxes or Tax Returns is outstanding, nor is any request for any such waiver or consent pending. There has been no Tax audit or other administrative proceeding or court proceeding with regard to any Taxes or Tax Returns, nor is any such Tax audit or other proceeding pending, nor has any notice to the Company in writing or otherwise to the knowledge of the Company from any taxing authority regarding any such Tax, audit or other proceeding, or, to the knowledge of the Company, is any such Tax audit or other proceeding threatened with regard to any Taxes or Tax Returns.
- (d) Neither the Company nor any Tax Affiliate is a party to any agreement, contract or arrangement that would result, separately or in the aggregate, in any payments in connection with Closing that would constitute "excess parachute payments" within the meaning of Section 280G of the Code.
- (e) Neither the Company nor any Tax Affiliate has requested or been granted any extension of time within which to file any Tax Return, which Tax Return has not since been filed.
- (f) The Company and any Tax Affiliate have made available to Parent in the Data Room copies of all federal and state income Tax Returns for all periods since January 1, 2005 and all state sales and use Tax Returns for all periods since January 1, 2007.
- (g) Neither the Company nor any Tax Affiliate has been a member of an affiliated group (within the meaning of Code Section 1504(a)) filing a consolidated federal income Tax Return or a combined or unitary state income Tax Return (other than a group the common parent of which was the Company). Neither the Company nor any Tax Affiliate is presently liable, nor does the Company or any Tax Affiliate have any potential liability, for the Taxes of

another person under Treasury Regulations Section 1.1502-6 (or comparable provision of state, local or foreign law), as transferee or successor, or by contract, indemnity or otherwise.

- (h) Neither the Company nor any Tax Affiliate is a party to any Tax sharing, indemnification or allocation agreement.
- (i) Neither the Company nor any Subsidiary constitutes either a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of shares qualifying for tax-free treatment under Section 355 of the Code (i) which took place during the two year period ending on the date of this Agreement or (ii) that could otherwise constitute part of a "plan" or "series of related transactions" (within the meaning of Section 355(e) of the Code) in conjunction with the Merger.
- (j) Neither the Company nor any Tax Affiliate has engaged in any transaction that is a listed transaction pursuant to Treasury Regulations Sections 1.6011-4.
- (k) No claim has ever been made, in writing or otherwise to the knowledge of the Company or any Subsidiary, by a taxing authority or other Governmental Entity in a jurisdiction where the Company or the Tax Affiliates do not file Tax Returns that they, individually or collectively, are or may be subject to taxation by that jurisdiction.
- (l) For purposes of this Agreement, the term "<u>Tax</u>" or "<u>Taxes</u>" means all taxes, charges, fees, levies, or other assessments, including all net income, gross income, gross receipts, sales, use, ad valorem, value-added tax (VAT), transfer, franchise, profits, license, withholding, payroll, employment, social security, unemployment, disability, workers' compensation, excise, estimated, severance, stamp, occupation, property, goods and services tax (GST) or other taxes, customs duties, premiums, contributions, fees, assessments, or charges, including all interest and penalties thereon, and additions to Tax or additional amounts in each case imposed by any taxing authority, domestic or foreign, upon either the Company or any Tax Affiliate (whether disputed or not). "<u>Tax Returns</u>" means all returns, declarations, reports, estimates, information returns, and statements required to be filed or sent in respect of any Taxes.

3.13 Contracts and Commitments.

- (a) Section 3.13 of the Disclosure Schedule lists the following contracts, commitments and/or binding understandings, whether oral or written, to which the Company or any Subsidiary is a party, which are in effect as of the date hereof (the "Contracts"):
 - (i) all employment or consulting agreements, all contracts or commitments providing for severance, termination or similar payments, including on a change of control of the Company, and all union, collective bargaining or similar agreements with labor representatives;
 - (ii) all distributor, reseller, OEM, dealer, manufacturer's representative, sales agency or advertising agency, finder's and manufacturing or assembly contracts;
 - (iii) all material contracts terminable by any other party thereto upon a change of control of the Company or any Subsidiary or upon the failure of the Company or any Subsidiary to satisfy financial or performance criteria specified in such contract as provided therein;
 - (iv) all leases of tangible personal property with aggregate annual payments in excess of \$100,000 per year;
 - (v) all contracts between or among the Company, any Subsidiary, any holder of Company Capital Stock or any affiliate of such holder, any director, officer or employee of the Company or any Subsidiary or any member of his or her immediate family or any entity affiliated with any such Person relating in any way to the Company or any Subsidiary (to the extent not otherwise disclosed in Section 3.21 of the Disclosure Schedule);

- (vi) all contracts relating to the performance and payment of any surety bond or letter of credit required to be maintained by the Company or any Subsidiary;
- (vii) all contracts obligating the Company, directly or indirectly, to guarantee the payment or performance of any other Person;
- (viii) all confidentiality or non-disclosure agreements, other than confidentiality or non-disclosure agreements executed in connection with the sale of products or services in the ordinary course of business;
- (ix) all agreements or indentures relating to the borrowing of money or otherwise placing a material mortgage, pledge or other Lien on any of the assets of the Company or any Subsidiary;
- (x) all contracts or group of related contracts with the same party for the purchase of products or services under which the undelivered balance of such products or services is in excess of \$250,000;
- (xi) all contracts or group of related contracts with the same party for the sale of products or services under which the undelivered balance of such products or services has a sales price in excess of \$500,000;
- (xii) all contracts containing exclusivity, noncompetition or nonsolicitation provisions or which would otherwise prohibit the Company or any Subsidiary from freely engaging in business anywhere in the world; and
- (xiii) all license agreements, transfer or joint-use agreements or other agreements providing for the payment or receipt of royalties, license, or similar fees by the Company or any Subsidiary on or after the date of signing of this Agreement in connection with the Company Intellectual Property (as defined in Section 3.14(a) hereof) except for (A) purchase, lease, maintenance and support agreements for computer systems, (B) so-called "off-the-shelf" and "shrink-wrap" license agreements for generally available software other than any such software that is included in any product of the Company or any Subsidiary or that is otherwise distributed by the Company or any Subsidiary, (C) agreements described in the Disclosure Schedules under Section 3.14(b), and (D) non-exclusive licenses to customers in connection with providing products and services in the ordinary course of business;
- (xiv) any and all other contracts or commitments for capital expenditures with an amount remaining to be paid in excess of \$100,000; and
- (xv) all agreements for the sale of any capital assets with an amount remaining to be paid in excess of \$100,000 (other than agreements for sales made by the Company in the ordinary course of business).
- (b) The Company or the applicable Subsidiary has performed in all material respects all obligations required to be performed by it as of the date hereof in connection with the Contracts and is not in receipt of any claim of default under any such Contract and, to the Company's knowledge, no such claim is threatened. Neither the Company nor any Subsidiary has a present expectation or intention of not fully performing any material obligation pursuant to any Contract. The Company has no knowledge of any breach or anticipated breach by any other party to any Contract. The Company has no knowledge that any existing contracts or subcontracts with the Company's or any Subsidiary's customers cannot be fully performed by the Company or the applicable Subsidiary. Neither the Company nor any Subsidiary has any obligation to refund payments received for work not yet performed under contracts where the percentage of work completed is less than the percentage of revenues received to date.
- (c) Prior to the date of this Agreement, the Company has made available to Parent in the Data Room a true and complete copy of each written Contract, and a written description of each oral Contract, together with all material amendments, waivers or other changes thereto.
 - 3.14 Intellectual Property Rights.

- (a) "<u>Intellectual Property</u>" means all patents, patent rights, trademarks, trademark rights, industrial designs, industrial design rights, trade names, trade name rights, service marks, service mark rights, domain names, copyrights, maskworks, and any applications for any of the foregoing, schematics, inventions, technology, know-how, trade secrets, ideas, algorithms, processes, computer software programs or applications (in both source code and object code form), database, and tangible or intangible proprietary or confidential information. Except as would not have a Material Adverse Effect, the Company, together with the Subsidiaries, owns, licenses or otherwise possesses rights to the Intellectual Property that is necessary to operate the business of the Company and the Subsidiaries as it is currently conducted and, to the knowledge of the Company, to operate the business of the Company and the Subsidiaries as it is currently proposed by the Company to be conducted (collectively, "<u>Company Intellectual Property</u>").
- (b) Section 3.14(b) of the Disclosure Schedule lists (i) all patents and patent applications and all registered trademarks and trademark applications, all registered industrial designs and industrial design applications, registered trade names, registered service marks and service mark applications, domain names and all registered copyrights owned by the Company or any Subsidiary, including the jurisdictions in which each such Intellectual Property right has been issued or registered or in which any application for such issuance and registration has been filed (the "Registered Intellectual Property"), (ii) all material licenses, sublicenses and other agreements other than nonexclusive licenses and sublicenses granted to customers in the ordinary course of business as to which the Company or any Subsidiary is a party and pursuant to which any third party is authorized to use any Registered Intellectual Property, and (iii) all material licenses, sublicenses and other agreements as to which the Company or any Subsidiary is a party and pursuant to which the Company or any Subsidiary is authorized to use any Intellectual Property of third parties ("Third Party Intellectual Property Rights") which are incorporated in, are, or form a part of or are distributed with any product of the Company or any Subsidiary. Neither the Company nor any Subsidiary is in material breach or violation of any such license, sublicense, or agreement, and, to the knowledge of the Company, no other party thereto is in, or is threatening, material breach or violation thereof, nor except as would not have a Material Adverse Effect. has the Company or any Subsidiary suffered any actual loss of rights relating to Company Intellectual Property thereunder. The Company or a Subsidiary is the sole and exclusive owner, with all right, title and interest in and to (free and clear of any Liens other than Permitted Liens), the Registered Intellectual Property, and has sole and exclusive rights (and is not contractually obligated to pay any compensation to any third party in respect thereof) to the use thereof in connection with the services or products in respect of which the Registered Intellectual Property is being used by the Company or such Subsidiary.
- (c) To the Company's knowledge, there is no unauthorized use, infringement or misappropriation of any Company or Subsidiary owned Intellectual Property, or any trade secret material of the Company or any Subsidiary, by any third party, including any current or former employee, contractor or independent consultant. Other than contractual obligations entered in the ordinary course of business in connection with the provision of the Company and its Subsidiaries' products and services, neither the Company nor any Subsidiary has entered into any agreement to indemnify any other Person against any charge of infringement of any Intellectual Property.
- (d) Neither the Company nor any Subsidiary has been sued in any suit, action or proceeding which involves, nor have any of them received a written request for indemnification or defense that relates to, a claim that the Company or any Subsidiary or that the making, using, or selling of any product or service of the Company or any Subsidiary has infringed, misappropriated, or violated any Intellectual Property of any third party. Moreover, (i) the conduct of the business of the Company and the Subsidiaries as currently conducted does not infringe, misappropriate, or violate any Intellectual Property of any third party; (ii) to the knowledge of the Company, the conduct of the business of the Company and the Subsidiaries as conducted in the past 4 years did not at the time infringe, misappropriate, or violate any Intellectual Property of any third party and the conduct of the business of the Company and the Subsidiaries as it is currently proposed by the Company to be conducted does not infringe, misappropriate, or violate any Intellectual Property of any third party; and (iii) to the actual knowledge of all current employees and contractors of the Company and the Subsidiaries, the conduct of the business of the Company and the Subsidiaries as currently conducted and currently proposed by the Company to be conducted will not infringe, misappropriate, or violate any Intellectual Property of any third party. To the knowledge of the Company, no third party is challenging the ownership by the

Company or any Subsidiary of any Company or Subsidiary owned Intellectual Property or the validity, enforceability or effectiveness thereof. Neither the Company nor any Subsidiary has brought any action, suit or proceeding for infringement of Company or Subsidiary owned Intellectual Property. There are no pending or, to the knowledge of the Company threatened, interference, re-examinations, oppositions or nullities involving any patents, patent rights or applications therefor of the Company or any Subsidiary, except such as may have been commenced by the Company or any Subsidiary.

- (e) Except as would not have a Material Adverse Effect, to the Company's knowledge, the Company and each Subsidiary have executed written agreements with their former and current employees and, as applicable, consultants, contractors and third parties involved in the creation of Intellectual Property on behalf of the Company or any Subsidiary which assign to the Company or such Subsidiary any and all rights to such Intellectual Property made by them during their service to the Company or such Subsidiary.
- (f) The Company and each Subsidiary have taken reasonably necessary and appropriate steps to protect and preserve the confidentiality in all material respects of their material trade secrets and other proprietary confidential information.
- (g) The Company, together with the Subsidiaries, solely owns all material Intellectual Property developed on behalf of the Company or its Subsidiaries by any third parties performing development, engineering, or manufacturing services on behalf of the Company.
- (h) None of the Company or any Subsidiary is a party to any agreement or arrangement, or, to the knowledge of the Company is subject to any law, rule, or regulation, under which any governmental entity had or has acquired rights, or has the right or ability to do so in the future, with respect to any product or service of the Company or any Subsidiary or any Company or Subsidiary owned Intellectual Property, except for non-exclusive licenses to customers in connection with providing products and services in the ordinary course of business.
- (i) None of the Company or any Subsidiary, or any employee, contractor, or consultant thereof, (i) has expressly granted or is obligated to grant, any right, title or interest in or to any of the Company or Subsidiary owned material Intellectual Property (including without limitation any obligation to license on a royalty-free or reasonable-and-non-discriminatory basis) pursuant to any agreement, by-law, understanding, or requirement of any standards or certification body, industry or governmental consortium, or trade group; or (ii) has submitted to any standards or certification body, industry or governmental consortium, or trade group any declaration, assurance, disclaimer, proposal, or other statement relating to any of the Company or Subsidiary owned material Intellectual Property.
- (i) No rights in material Company Software have been transferred by the Company or any Subsidiary to any third party except to customers of the Company or any Subsidiary to whom it has licensed such Company Software in object code form in the ordinary course of business on a non-exclusive basis without the right to sublicense or make derivative works. The Company or any Subsidiary, as the case may be, has the right to use all software development tools, library functions, compilers, and other third party software that it uses to operate, modify, distribute, or support the Company Software. Except as would not have a Material Adverse Effect, none of the Company or any Subsidiary has provided, escrowed, or otherwise disclosed the source code for any Company Software to any third party nor is it under any obligation to do so. Except as would not have a Material Adverse Effect, none of the Company or any Subsidiary has (i) incorporated any Publicly Available Software in whole or in part into any part of Company Software in a manner that subjects the Company Software, in whole or in part, to all or part of the license obligations of any Publicly Available Software or that grants to any third party, any rights or immunities under the Company or Subsidiary owned Intellectual Property; (ii) used Publicly Available Software in whole or in part in the development of the Company Software in a manner that subjects the Company Software, in whole or in part, to all or part of the license obligations of any Publicly Available Software or that grants to any third party, any rights or immunities under the Company or Subsidiary owned Intellectual Property; or (iii) distributed Company Software in conjunction with or for use with any Publicly Available Software in a manner that subjects the Company Software, in whole or in part, to

all or part of the license obligations of any Publicly Available Software or that grants to any third party, any rights or immunities under the Company or Subsidiary owned Intellectual Property. For purposes of this Agreement, "Company Software" means Company or Subsidiary owned software included in any product of the Company or any Subsidiary or that is otherwise distributed by the Company or any Subsidiary (but does not include any Publicly Available Software or other software in-licensed from third parties). For the purposes of this Agreement "Publicly Available Software" means each of: (a) any software that is distributed as free software, open source software or similar licensing or distribution models by a third party; and (b) any software that requires as a condition of use, modification and/or distribution of such software that such software or other software incorporated into, derived from or distributed with such software (i) be disclosed or distributed in source code form, (ii) be licensed for the purpose of making derivative works, or (iii) be redistributable at no charge.

- 3.15 <u>Litigation</u>. There are no actions, arbitrations, mediations, suits, proceedings, orders, or to the knowledge of the Company, investigations pending against the Company or any Subsidiary, at law or in equity, before or by any federal, state, municipal or other governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign. To the knowledge of the Company, there are no such actions, arbitrations, mediations, suits, proceedings, orders or investigations threatened against the Company or any Subsidiary.
- 3.16 Warranties. Section 3.16 of the Disclosure Schedule summarizes all claims outstanding, pending or, to the knowledge of the Company, threatened for breach of any warranty relating to products sold or services provided by the Company or any Subsidiary prior to the date hereof. The amounts the Company reasonably expects to be required to settle such claims in the aggregate are not in excess of the reserve for product or service warranty claims set forth on the face of the Latest Balance Sheet. Descriptions of the Company's or any Subsidiary's product and/or service warranties have been made available to Parent in the Data Room and are correct and complete in all material respects. The Company's products have been manufactured in compliance in all material respects with (i) the specifications of such products as set forth in any published marketing, advertising or other descriptive materials which the Company makes available or has made available to its customers and (ii) with respect to any products manufactured to the specifications of a customer pursuant to a written contract with such customer, the specifications of such customer.

3.17 Employees.

- (a) To the knowledge of the Company, no executive employee of the Company or any Subsidiary and no group of the Company's or any Subsidiary's employees has any plans to terminate his, her or their employment. The Company and each Subsidiary has complied in all material respects at all times with all laws relating to the employment of labor, including provisions thereof relating to wages, hours, equal opportunity, collective bargaining and the payment of social security and other employment-related Taxes. There are no workers' compensation claims pending against the Company, or any Subsidiary. To the knowledge of the Company, no employee of the Company or any Subsidiary is subject to any secrecy or noncompetition agreement or any other agreement or restriction of any kind that would impede in any way the ability of such employee to carry out fully all activities of such employee in furtherance of the business of the Company and each Subsidiary.
- (b) The employment of any terminated former employee of the Company or any Subsidiary has been terminated in accordance with any applicable contractual terms and applicable law, and the Company and the Subsidiaries do not have any material liability under any contract or applicable law toward any such terminated employee. The consummation of the Merger or the other transactions contemplated hereby will not, in and of itself, cause the Company or any Subsidiary to incur or suffer any liability relating to, or obligation to pay, severance, termination or other payments to any Person.
- (c) Neither the Company nor any Subsidiary has made any loans (except advances against accrued salaries or for business travel, lodging or other expenses in the normal course of business) to any employee of the Company or any Subsidiary.

- (d) There are no strikes, slowdowns, work stoppages or other material labor controversies pending or, to the knowledge of the Company, threatened in writing against or otherwise affecting the employees or facilities of the Company or any Subsidiary. Neither the Company nor any Subsidiary has experienced any labor strike, slowdown, work stoppage exceeding 24 hours or other material labor controversy involving its employees within the past two years. No labor representatives hold bargaining rights with respect to any employees of the Company or any Subsidiary, and there are, to the knowledge of the Company, no current or threatened attempts to organize or establish any trade union or employee association with respect to the Company or any Subsidiary. The Company and each Subsidiary has paid in full to all employees all wages, salaries and commissions due and payable to such employees and has fully reserved on such Company's or such Subsidiary's accounts all amounts for wages, salaries and commissions due but not yet payable to such employees. There have been no lay-offs of employees or senior employees or work reduction programs undertaken by or on behalf of the Company or the Subsidiaries in the past two years, and no such programs have been adopted by any of the Company's or Subsidiaries' boards of directors or publicly announced.
- (e) Section 3.17(e) of the Disclosure Schedule sets forth a list of all employees of the Company who hold a temporary work authorization, including H-1B, F-1 or J-1 visas or work authorizations (the "*Work Permits*"), setting forth the name of the employees, the type of Work Permit and the length of time remaining on such Work Permit. The Company has not received any notice from the US Citizenship and Immigration Services, a bureau of the Department of Homeland Security (the "*USCIS*") or any other Governmental Entity that any Work Permit has been revoked. To the knowledge of the Company, there is no action pending or threatened in writing to revoke or adversely modify the terms of any of the Work Permits.

3.18 Employee Benefit Plans.

- (a) "ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and "Plan" means every plan, fund, contract, program and arrangement (whether written or not) which is maintained or contributed to by the Company or any ERISA Affiliate for the benefit of present or former employees or with respect to which the Company otherwise has any liability for premiums or benefits. "Plan" includes any arrangement intended to provide: (i) medical, surgical, health care, hospitalization, dental, vision, workers' compensation, life insurance, death, disability, legal services, severance, sickness, accident, or cafeteria plan benefits (whether or not defined in Section 3(1) of ERISA), (ii) pension, profit sharing, stock bonus, retirement, supplemental retirement or deferred compensation benefits (whether or not Tax qualified and whether or not defined in Section 3(2) of ERISA), (iii) bonus, incentive compensation, stock option, stock appreciation right, phantom stock or stock purchase benefits, change in control benefits or (iv) salary continuation, termination pay, vacation or holiday benefits (whether or not defined in Section 3(3) of ERISA). Section 3.18(a) of the Disclosure Schedule sets forth a list of all Plans. "ERISA Affiliate" means each Subsidiary and any trade or business (whether or not incorporated) that is part of the same controlled group under, common control with, or part of an affiliated service group that includes, the Company, within the meaning of Code Section 414(b), (c), (m) or (o).
- (b) To the knowledge of the Company or any ERISA Affiliate, and except as would not reasonably be expected to have a Material Adverse Effect, (i) each Plan complies in all respects with the requirements of ERISA and the Code (including those necessary to obtain intended tax treatment and tax benefits); (ii) all required contributions which are due have been made and an accrual to the extent required by GAAP has been made on the books and records of the Company or any ERISA Affiliate; (iii) there are no actions, suits or claims pending, other than routine uncontested claims for benefits; and (iv) there have been no nonexempt prohibited transactions (as defined in Section 406 of ERISA or Section 4975 of the Code).
- (c) Parent has received or has had made available in the Data Room to it true and complete copies of (i) all Plan documents, including related trust agreements or funding arrangements; (ii) the most recent determination letter, if any, received by the Company or any ERISA Affiliate from the Internal Revenue Service (the "<u>IRS</u>") regarding the Plans and any amendment to any Plan made subsequent to any Plan amendments covered by any such determination

- letter; (iii) the most recent financial statements for the Plans, if any; (iv) the most recently prepared actuarial valuation reports, if any; (v) current summary plan descriptions and summaries of material modification; (vi) annual returns/ reports on Form 5500 and summary annual reports for each of the most recent three plan years, (vii) any filings with the IRS or the U.S. Department of Labor within the last five years preceding the date of this Agreement, and (viii) insurance contracts, service agreements, and related documents. To the knowledge of the Company or any ERISA Affiliate, nothing has occurred that could materially adversely affect the qualification of the Plans and their related trusts.
- (d) Neither the Company nor any ERISA Affiliate maintains or contributes to any multiemployer plan, as defined in Section 3(37) of ERISA. Neither the Company nor any ERISA Affiliate has any material liabilities under Title IV of ERISA, including under Section 4201 of ERISA for any complete or partial withdrawal from a multiemployer plan.
- (e) No Plan provides benefits or coverage in the nature of health or life insurance after separation from employment, other than (i) death benefits under the employee benefit plans or programs (whether or not subject to ERISA) set forth in Section 3.16(e) of the Disclosure Schedule and (ii) health care continuation benefits described in Section 4980B of the Code or other applicable law.
- (f) To the knowledge of the Company or any ERISA Affiliate, the events contemplated by this Agreement will not trigger, accelerate, or otherwise entitle any current or former employees of the Company or any ERISA Affiliate to severance or other benefits.
- (g) To the knowledge of the Company or any ERISA Affiliate, neither the Company, an ERISA Affiliate nor any of their respective directors, officers, employees or other "fiduciaries", as such term is defined in Section 3(21) of ERISA, has committed any material breach of fiduciary responsibility imposed by ERISA or any other applicable law with respect to the Plans which would subject the Company, an ERISA Affiliate, Parent or any of their respective directors, officers or employees to any material liability under ERISA or any applicable law.
- (h) There are no (i) legal, administrative or other proceedings or governmental investigations or audits, or (ii) complaints to or by any Governmental Entity, which are pending, anticipated or, to the knowledge of the Company or any ERISA Affiliate, threatened, against any Plan or its assets.
- (i) There are no leased employees, as defined in Section 414(n) of the Code, providing services to the Company or any ERISA Affiliate, that must be taken into account with respect to the requirements under Section 414(n)(3) of the Code.
- (j) No provision of any Plan could reasonably be expected to result in any limitation on the Company, an ERISA Affiliate or Parent from amending or terminating any Plan.
- (k) All Nonqualified Deferred Compensation Plans (as defined in Code Section 409A(d)(1)) of the Company or any ERISA Affiliate will be operated in material compliance with Code Section 409A on or before the Effective Date and neither those Plans nor this transaction will cause a participant in such Plans to be subject to the Tax imposed by Code Section 409A(a)(1)(B).
 - 3.19 Pension, Benefits Plans (Non-U.S. Operations). With respect to its operations outside of the United States:
- (a) Section 3.19(a) of the Disclosure Schedule lists every material retirement, pension, incentive compensation, deferred compensation, profit-sharing, stock purchase or stock option plan; multi-employer pension plan; life, health, disability or other insurance plan; severance or separation plan; or any other employees benefit plan, practice or arrangement of any kind, whether written or oral (collectively, "*International Employee Benefit Plans*") maintained by the Company or an ERISA Affiliate or to which the Company contributes or is subject with respect to all of its employees and former employees and all dependents and beneficiaries of such employees and former employees.

- (b) Other than the International Employee Benefit Plans and any statutory employee benefits, there are no material plans or arrangements under which benefits of any kind are or may be payable by it to its employees upon retirement or death, or in the event of disability or sickness or other circumstances.
- (c) The International Employee Benefit Plans have at all times materially complied with and been duly administered in accordance with, and will between the date hereof and the Closing materially comply with and be duly administered in accordance with, the terms of such Plans and with all applicable laws and regulations and requirements having force of law and will not, without the consent of the other party hereto, be amended or discontinued (other than in the ordinary course of business or as required by law) before the Closing.
- 3.20 <u>Insurance</u>. Each of the Company and the Subsidiaries maintains insurance relating to its business and covering property, fire, casualty, liability, life and workers' compensation. Such insurance: (a) is in full force and effect; (b) is sufficient for compliance, in all material respects, with all requirements of applicable law and of any contract or agreement to which the Company or such Subsidiary is subject; and (c) provides commercially reasonable insurance coverage for the activities of the Company or such Subsidiary.
- 3.21 Affiliate Transactions. Other than pursuant to this Agreement, no officer or director of the Company or any Subsidiary or any member of the immediate family of any such officer or director, or any entity in which any of such Persons owns any beneficial interest (other than any publicly held corporation whose stock is traded on a national securities exchange or in the over-the-counter market and less than five percent of the stock of which is beneficially owned by any of such Persons) (collectively "Insiders"), has any agreement with the Company or any Subsidiary (other than employment arrangements) or any interest in any property, real, personal or mixed, tangible or intangible, used in the business of the Company or any Subsidiary (other than ownership of capital stock or securities to acquire capital stock of the Company). None of the Insiders has any direct or indirect interest (other than any publicly held corporation whose stock is traded on a national securities exchange or in the over-the-counter market and less than two percent of the stock of which is beneficially owned by any of such Persons) in any competitor, vendor, supplier or customer of the Company or any Subsidiary or in any Person from whom or to whom the Company or any Subsidiary leases any property, or in any other Person with whom the Company or any Subsidiary transacts business of any material nature. For purposes of this Section 3.21, the members of the immediate family of an officer or director shall consist of the spouse, parents, children, siblings, mothers- and fathers-in-law, sons- and daughters-in-law, and brothers- and sisters-in-law of such officer, director or employee.
- 3.22 <u>Customers and Supplier</u>. Section 3.22 of the Disclosure Schedule lists the 10 largest customers and the 10 largest suppliers of the Company and the Subsidiaries (on a consolidated basis) for the fiscal year ended December 31, 2006 and for the six-month period ended June 30, 2007 and sets forth opposite the name of each such customer or supplier the approximate percentage of consolidated net sales or purchases by the Company and the Subsidiaries attributable to such customer or supplier for each such period. Since the Balance Sheet Date, no customer or supplier listed in Section 3.22 of the Disclosure Schedule has notified the Company or any of its Subsidiaries that it will stop or substantially decrease the rate of business done with the Company and the Subsidiaries, taken as a whole, except for changes in the ordinary course of the Company's or the Subsidiaries' business.

3.23 Compliance with Laws; Permits.

(a) The Company and each Subsidiary and each of their respective officers, directors, agents and employees have complied in all material respects with all applicable laws, including, but not limited to, federal, state, local and foreign laws pertaining to product labeling, consumer products safety, equal employment opportunity, immigration, employee retirement, affirmative action and other hiring practices, occupational safety and health, workers' compensation, unemployment and building and zoning codes, pertaining to the business, the assets of the Company or the Subsidiaries or the Real Property and to which the Company or the Subsidiaries may be subject and no claims have been filed against the Company or any Subsidiary alleging a violation of any such laws. Neither the Company nor any

Subsidiary is relying on any exemption from or deferral of any such applicable law that would not be available to it immediately after the Closing.

- (b) The Company and each Subsidiary have, in full force and effect, all material licenses, approvals, permits and certificates, from federal, state and foreign authorities and all material licenses, approvals, permits and certificates from local authorities, in each case, necessary to conduct their respective business and own and operate their respective assets in all material respects (collectively, the "*Permits*"). The Company and each Subsidiary have conducted their business in material compliance with all material terms and conditions of the Permits.
- (c) Neither the Company, any Subsidiary nor to the Company's knowledge any Person representing the Company or any Subsidiary has offered, authorized, promised, made or agreed to make gifts of money, other property or similar benefits (other than incidental gifts of articles of nominal value) to any actual or potential customer, supplier, governmental employee, political party, political party official or candidate, official of a public international organization or any other Person in a position to assist or hinder the Company or the Subsidiaries in connection with any actual or proposed transaction, other than payments required or permitted by the laws of the applicable jurisdiction; *provided*, *however*, that any such payments shall have been in compliance with the U.S. Foreign Corrupt Practices Act.
- (d) The Company and the Subsidiaries have conducted their export transactions in accordance with applicable provisions of export control laws, including any laws relating to the export of technology.
- (e) Neither the Company nor any Subsidiary has and, neither has had in the past, any legal obligation to file any form, report, schedule, statement, proxy statement or other document or information with the SEC, and has not filed with the SEC any such form, report, schedule, statement, proxy statement or other document or information.

3.24 Environmental Matters.

- (a) As used in this Section 3.24, the following terms have the following meanings:
- (i) "Environmental Costs" means any and all costs and expenditures, including reasonable attorney's fees and any fees of environmental consultants or engineers incurred in connection with investigating, defending, remediating or otherwise responding to any Release of Hazardous Materials, any violation or alleged violation of Environmental Law, any fees, fines, penalties or charges associated with any governmental authorization, or any actions necessary to comply with any Environmental Law.
- (ii) "Environmental Laws" means all applicable federal, state, local, foreign and other laws, orders, decrees, directives, permits, licenses and judgments relating to pollution, contamination or protection of the environment (including all applicable federal, state, local and other laws, orders, decrees, directives, permits, licenses and judgments relating to Hazardous Materials in effect as of the date of this Agreement) including the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended ("CERCLA"), and the Waste of Electronic and Electrical Equipment and Reduction of Hazardous Substances Directives of the European Union
- (iii) "*Hazardous Materials*" means any dangerous, toxic or hazardous pollutant, contaminant, chemical, waste, material or substance as defined in or governed by any federal, state, local, foreign or other law or other requirement relating to such substance or otherwise relating to the environment or human health or safety, including any waste, material, substance, pollutant or contaminant that might cause any injury to human health or safety or to the environment or might subject the Company or any Subsidiary to any imposition of costs or liability under any Environmental Law.

- (iv) "<u>List</u>" means the United States Environmental Protection Agency's National Priorities List of Hazardous Waste Sites promulgated under CERCLA or any state remedial priority list promulgated or published pursuant to any comparable California law.
- (v) "<u>Regulatory Action</u>" means any litigation with respect to the Company or any Subsidiary brought or instigated by any Governmental Entity in connection with any Environmental Costs, Release of Hazardous Materials or any Environmental Law.
- (vi) "*Release*" means the spilling, leaking, disposing, discharging, emitting, depositing, ejecting, leaching, escaping or any other release, however defined, whether intentional or unintentional, of any Hazardous Material.
- (vii) "<u>Third-Party Environmental Claim</u>" means any Litigation (other than a Regulatory Action) based on negligence, trespass, strict liability, nuisance, toxic tort or any other cause of action or theory relating to any Environmental Costs, Release of Hazardous Materials or any violation of Environmental Law.
- (b) No Third-Party Environmental Claim or Regulatory Action is pending or, to the knowledge of the Company, threatened against the Company or any Subsidiary, which would, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
 - (c) No Real Property is listed on a List.
- (d) All transportation of Hazardous Materials by the Company or any Subsidiary to properties not owned, leased or operated by the Company or any Subsidiary has been in compliance, in all material respects, with applicable Environmental Law. The Company has not transported or arranged for the transportation of any Hazardous Materials to any location that is (i) listed on the List or (ii) to the knowledge of the Company, the subject of any material Regulatory Action or Third-Party Environmental Claim.
- (e) To the knowledge of the Company, no Real Property has ever been used as a landfill, dump or other disposal, storage, transfer, handling or treatment area for Hazardous Materials, or as a gasoline service station or a facility for selling, dispensing, storing, transferring, disposing or handling petroleum and/or petroleum products.
- (f) To the knowledge of the Company, there has not been any Release of any Hazardous Material on, under, about, or from the Real Property which would, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- (g) The Company's and the Subsidiary's use and operation of the Real Property is and has been at all times in compliance, in all material respects, with all applicable Environmental Law.
- (h) Neither the Company nor any Subsidiary has generated, treated, contained, handled, located, used, manufactured, processed, buried, incinerated, deposited or stored any Hazardous Materials on or about any part of the Real Property except in compliance, in all material respects, with all Environmental Laws.
- (i) No aboveground or, to the knowledge of the Company, underground storage tanks are located on, under or about the Real Property.
- (j) To the knowledge of the Company, all material environmental reports and investigations in the Company's possession with respect to the Company, any Subsidiary or the Real Property are listed on Section 3.24(j) of the Disclosure Schedule.
- (k) To the knowledge of the Company, no encumbrance has been attached or filed against the Company or any Subsidiary in favor of any Person for (i) any liability under or violation of any applicable Environmental Law, (ii) any Release of Hazardous Materials or (iii) any imposition of Environmental Costs.

- 3.25 <u>Bank Accounts</u>. Section 3.25 of the Disclosure Schedule sets forth a full and complete list of all bank accounts and safe deposit boxes of the Company and each Subsidiary, the number of each such account or box, and the names of the Persons authorized to draw on such accounts or to access such boxes. All cash in such accounts is held in demand deposits and is not subject to any restriction or documentation as to withdrawal, except for restrictions relating to authorized signatories or other restrictions imposed by the bank pursuant to its policies and procedures.
- 3.26 <u>Vote Required</u>. The only votes of the holders of any class or series of Company Capital Stock necessary to approve this Agreement, the Merger and the transactions contemplated by this Agreement are (i) the affirmative vote of the holders of a majority of the outstanding shares of Series A-1 Preferred Stock, voting as a single class, (ii) the affirmative vote of the holders of a majority of the outstanding shares of Series B-1 Preferred Stock, voting as a single class, (iii) the affirmative election of the holders of 66 2/3% of the then outstanding shares of Series A-1 Preferred Stock to convert the Series A-1 Preferred Stock into Company Common Stock, (iv) the affirmative vote of the holders of a majority of the outstanding shares of Series A-1 Preferred Stock, Series B-1 Preferred Stock and Company Common Stock, voting as a single class (assuming that each share of Series A-1 Preferred Stock and each share of Series B-1 Preferred Stock has been converted to shares of Company Common Stock in accordance with their respective terms) and (v) the affirmative vote of the holders of a majority of the outstanding shares of Series A-1 Preferred Stock and Company Common Stock, voting as a single class, and no other vote of the holders of any class or series of the capital stock of the Company is necessary to approve the other transactions contemplated hereby.

3.27 Anti-takeover Plan; State Takeover Statutes.

- (a) Neither the Company nor any Subsidiary of the Company has in effect any stockholder rights plan or similar device or arrangement, commonly or colloquially known as a "poison pill" or "anti-takeover" plan, or any similar plan, device or arrangement, and the board of directors of the Company has not adopted or authorized the adoption of such a plan, device or arrangement.
- (b) No "moratorium," "control share acquisition," "business combination," "fair price" or other form of antitakeover laws and regulations applies or purports to apply to the Merger, this Agreement or any of the transactions contemplated by this Agreement.
- 3.28 <u>Indemnification Obligations</u>. The Company has no knowledge of any action, proceeding or other event pending or threatened against any officer or director of the Company or any Subsidiary which would give rise to any indemnification obligation of the Company or any Subsidiary to its officers and directors under its Charter Documents, Governing Documents or any agreement between the Company or any Subsidiary and any of its officers or directors.
- 3.29 <u>Brokerage</u>. No third party shall be entitled to receive any brokerage commissions, finder's fees, fees for financial advisory services or other transaction fees customarily paid to financial advisors or brokers in connection with the transactions contemplated by this Agreement based on any arrangement or agreement made by or on behalf of the Company.

Article IV

Representations and Warranties of Parent

Parent hereby represents and warrants to the Company that:

4.1 <u>Incorporation and Corporate Power</u>. Parent is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Minnesota, with the requisite corporate power and authority to enter into this Agreement and perform its obligations hereunder. Merger Sub is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Delaware, with the requisite corporate power and authority to enter into this Agreement and perform its obligations hereunder. Merger Sub is duly qualified as a foreign corporation to do

business in every jurisdiction in which the nature of its business or its ownership of property requires it to be so qualified. Neither Parent nor Merger Sub is in violation of any of its respective Governing Documents.

- 4.2 Execution, Delivery; Valid and Binding Agreement. Each of Parent and Merger Sub has all requisite corporate power and authority to execute and deliver, and perform its obligations under, this Agreement and to consummate the transactions contemplated hereby. The execution, delivery and performance of this Agreement by each of Parent and Merger Sub, and the consummation of the transactions contemplated hereby, have been duly and validly authorized by all requisite corporate and shareholder or stockholder action, respectively, and no other corporate proceedings or shareholder or stockholder approvals, respectively, on its part are necessary to authorize the execution, delivery or performance of this Agreement. This Agreement has been duly and validly executed and delivered by each of Parent and Merger Sub and constitutes the valid and binding obligation of Parent and Merger Sub, respectively, enforceable against Parent and Merger Sub in accordance with its terms, and the other documents contemplated hereby, when executed and delivered by Parent, will constitute the valid and binding obligations of Parent, enforceable against Parent in accordance with their respective terms, in each case except to the extent that their enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other laws affecting the enforcement of creditors' rights generally and by general equitable principles.
- 4.3 No Violations, etc. The execution, delivery and performance of this Agreement by Parent and Merger Sub does not and the consummation of the transactions contemplated hereby will not: (a) contravene any provision of the Governing Documents of Parent or Merger Sub; (b) violate or conflict in any material respect with any federal, state, local or foreign law or any decree, writ, injunction, judgment or order of any court or administrative or other governmental body or of any arbitration award which is either applicable to, binding upon or enforceable against Parent or Merger Sub, or the business or any assets of Parent or Merger Sub; (c) conflict with or result in any material breach of any of the provisions of, or constitute a material default (or any event which would, with the passage of time or the giving of notice or both, constitute a material default) under, result in a material violation of, result in the creation of a right of termination, amendment, modification, abandonment or acceleration under any indenture, hypothecation, mortgage, lease, license, loan agreement or other material agreement or instrument which is either binding upon or enforceable against Parent or Merger Sub; (d) result in the creation of any material Lien upon Parent or Merger Sub or any of the assets of Parent or Merger Sub; or (e) require any authorization, consent, approval, exemption or other action by or notice to any Governmental Entity or any other third party, *other than* in connection with the applicable requirements of the HSR Act and any other comparable foreign merger or competition laws listed in Section 3.5 of the Disclosure Schedule.
- 4.4 <u>Litigation</u>. Except as disclosed in registration statements, prospectuses, forms, reports, schedules, statements and other documents (including exhibits and all other information incorporated by reference) filed or required to be filed by Parent with the SEC since January 1, 2006 to the date hereof, there are no material actions, arbitrations, mediations, suits, proceedings, orders or investigations pending or threatened in writing against Parent, at law or in equity, or before or by any federal, state, municipal or other governmental department, commission, board bureau, agency or instrumentality, domestic or foreign or any other Person.
- 4.5 <u>Financing</u>. Parent has, or will have at Closing, funds sufficient to consummate the transactions contemplated by this Agreement.
- 4.6 <u>Parent Rollover Option and Common Stock</u>. Parent has validly reserved a sufficient number of authorized but unissued shares of common stock to be issued by Parent upon exercise of the Parent Rollover Options. The Parent Rollover Options and the shares of common stock of Parent to be issued pursuant to the exercise of the Parent Rollover Option will be duly and validly authorized and issued, will be issued in compliance with all federal and state securities and state blue sky laws, will not be issued in violation of the preemptive rights of any stockholder of Parent and shall be free of any liens or encumbrances and the common stock of Parent when issued upon exercise of the Parent Rollover Options will be fully paid and non-assessable.

- 4.7 SEC Reports; Financial Statements. Since January 1, 2006, Parent has filed all required reports, schedules, forms, statements and other documents (including exhibits and all other information incorporated therein) with the SEC (the "Parent SEC Documents"). As of their respective dates, the Parent SEC Documents complied in all material respects with the requirements of the Securities Act or the Exchange Act, as the case may be, and the rules and regulations of the SEC promulgated thereunder applicable to such Parent SEC Documents and, except to the extent that information contained in any Parent SEC Document has been revised, modified or superseded by a later filed Parent SEC Document, none of the Parent SEC Documents when filed contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein. in light of the circumstances under which they were made, not misleading. The financial statements of Parent included in the Parent SEC Documents comply as to form, as of their respective dates of filing with the SEC, in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto, have been prepared in accordance with GAAP (except, in the case of unaudited statements, as otherwise permitted by rules and regulations applicable to the filing of a Quarterly Report on Form 10-Q under the Exchange Act) applied on a consistent basis during the periods involved (except as may be indicated in the notes thereto) and fairly present in all material respects the consolidated financial position of Parent and its consolidated Subsidiaries as of the dates thereof and the consolidated results of their operations and cash flows for the periods then ended (subject, in the case of unaudited statements, to normal recurring year-end audit adjustments).
- 4.8 <u>Reliance</u>. Except as otherwise set forth in this Agreement or in any agreement delivered at the Closing pursuant to Section 7.2, Parent has not relied on and is not relying on any representations, warranties or other assurances regarding the Company or its business other than those representations and warranties expressly set forth in this Agreement or any document delivered at Closing.

Article V

Conduct Prior to the Closing

- 5.1 <u>Conduct of the Business</u>. The Company shall, and shall cause each Subsidiary to, observe each term set forth in this Section 5.1 and agrees that, except pursuant to Contracts in existence as of the date hereof which have been made available to Parent in the Data Room and except as set forth in *Schedule 5.1* hereto, from the date hereof until the Closing Date, unless otherwise consented to by Parent in writing, which consent shall not be unreasonably withheld, delayed or conditioned:
- (a) The business of the Company and each Subsidiary shall be conducted only in, and none of the Company or the Subsidiaries shall take any action except in the ordinary course of the Company's and each Subsidiary's business and in accordance in all material respects with all applicable laws and the Company's and each Subsidiary's past custom and practice;
- (b) Neither the Company nor any Subsidiary shall, directly or indirectly, do or permit to occur any of the following: (i) issue or sell any additional shares of, or any options, warrants, conversion privileges or rights of any kind to acquire any shares of its capital stock except for issuances pursuant to the exercise or conversion of Company Options, Company Warrants and Convertible Notes outstanding on the date hereof, each as listed in Section 3.3 of the Disclosure Schedule; (ii) sell, pledge, dispose of or encumber any of its assets, except in the ordinary course of business; (iii) amend or propose to amend its Charter Documents or Governing Documents; (iv) split, combine or reclassify any outstanding shares of its capital stock, or declare, set aside or pay any dividend or other distribution payable in cash, stock, property or otherwise with respect to its capital stock; (v) redeem, purchase or acquire or offer to acquire any shares of its capital stock or other securities of such entity; (vi) acquire (by merger, exchange, consolidation, acquisition of stock or assets or otherwise) any interest in any corporation, partnership, joint venture or other business organization or division or material assets thereof; (vii) incur any indebtedness for borrowed money (excluding advances on existing credit facilities) or issue any debt securities; (viii) permit any accounts payable owed to trade creditors to remain outstanding more than 90 days except for (A) accounts payable where the terms or the

customary course of dealing that the Company or Subsidiary has with such creditors allows the accounts payable owed to such creditor to remain outstanding more than 90 days and (B) accounts payable with vendors outstanding more than 90 days which do not exceed \$20,000 individually and \$100,000 in the aggregate; (ix) accelerate, beyond the normal collection cycle, collection of accounts receivable; (x) materially amend or modify any Contract listed on Section 3.13 of the Disclosure Schedule, except in the normal course of business, or enter into any contract or agreement that otherwise would be listed on Section 3.13 of the Disclosure Schedule if such contract or agreement had been in effect on the date of this Agreement; or (xi) enter into or propose to enter into, or modify or propose to modify, any agreement, arrangement or understanding with respect to any of the matters set forth in this Section 5.1(b);

- (c) Other than as specifically contemplated by this Agreement or as required by law, neither the Company nor any Subsidiary shall, directly or indirectly, (i) enter into or modify any employment, severance or similar agreements or arrangements with, or grant any bonuses, salary increases, severance or termination pay to, any officers, directors or consultants, or (ii) in the case of employees who are not officers, take any action with respect to the grant of any bonuses, salary increases, severance or termination pay or with respect to any increase of benefits payable in effect on the date hereof;
- (d) Other than as specifically contemplated by this Agreement or as required by law, neither the Company nor any Subsidiary shall adopt or amend any bonus, profit sharing, compensation, stock option, pension, retirement, deferred compensation, employment or other employee benefit plan, trust, fund or group arrangement for the benefit or welfare of any employees, or any bonus, profit sharing, compensation, stock option, pension, retirement, deferred compensation, employment or other employee benefit plan, agreement, trust, fund or arrangements for the benefit or welfare of any director;
- (e) Neither the Company nor any Subsidiary shall cancel or terminate its current insurance policies or cause any of the coverage thereunder to lapse, unless simultaneously with such termination, cancellation or lapse, replacement policies providing coverage equal to or greater than the coverage under the canceled, terminated or lapsed policies for substantially similar premiums are in full force and effect;
- (f) The Company and each Subsidiary shall use its best efforts to (i) preserve intact the Company's and each Subsidiary's business organization and goodwill, keep available the services of the Company's and each Subsidiary's officers and employees as a group and maintain satisfactory relationships with suppliers, distributors, customers and others having business relationships with the Company and each Subsidiary; (ii) confer on a reasonable basis with representatives of Parent to report on operational matters and the general status of ongoing operations; (iii) not intentionally take any action which would render, or which reasonably may be expected to render, any representation or warranty made by it in this Agreement untrue at the Closing; (iv) notify Parent of any emergency or other change in the normal course of the Company's and each Subsidiary's business or in the operation of the Company's and each Subsidiary's properties and of any governmental or third party complaints, investigations or hearings (or communications indicating that the same may be contemplated); and (v) promptly notify Parent in writing if the Company or any Subsidiary shall discover that any representation or warranty made by it in this Agreement was when made, or has subsequently become, untrue in any material respect;
- (g) The Company (for purposes of this Section 5.1(g), all references to the Company shall include the Tax Affiliates) shall file (or cause to be filed) at its own expense, on or prior to the due date, all Tax Returns, including all Tax Returns and reports relating to the Plans, for all Tax periods ending on or before the Closing Date where the due date for such Tax Returns or reports (taking into account valid extensions of the respective due dates) falls on or before the Closing Date (all Tax Returns described in this Section 5.1(g) and any schedules to be included therewith shall be, except as required by law, prepared on a basis consistent with the Tax Returns of the Company prepared for prior Tax periods); *provided*, *however*, that the Company shall provide Parent with a copy of each federal and state income or franchise Tax Return (but excluding, for the avoidance of doubt, Tax Returns not related to income or franchise Taxes, such as, without limitation, Tax Returns related to withholding or employment Taxes) at least ten (10) days prior to the due date for filing such Tax Return (including extensions) for Parent's review, comment and

approval, which approval shall not be unreasonably withheld or delayed, and the Company shall reasonably cooperate with any request of Parent in connection with such Tax Returns. The Company and Parent shall use their commercially reasonable best efforts to resolve any disputes related to such Tax Returns. In the event that the Company and Parent are not able to resolve a dispute with respect to such Tax Returns by the fifth day prior to the due date for such Tax Returns, such dispute shall be submitted for resolution to an independent accounting firm mutually agreed upon by Company and Parent. The determination made by the independent accounting firm shall be final and binding. Company and Parent shall share equally the costs of such independent accounting firm.

- (h) Neither the Company nor any Subsidiary shall (i) make or rescind any material election relating to Taxes, (ii) amend any material Tax Return, (iii) consent to any adjustment of or settle or compromise, in each case, any material claim, action, suit, litigation, proceeding, arbitration, investigation or audit relating to Taxes, or (iv) agree to the waiver or any extension of the statute of limitations relating to any Taxes of the Company or any Tax Affiliate for any period that ends on or before the Closing Date; Company shall notify Parent within five (5) days of receipt of notice by Company of any Tax audit or Tax investigation by any taxing authority, and shall provide Parent with copies of all correspondence related to such Tax audit or investigation.
- (i) The Company and the Subsidiaries shall not change any of their Tax or financial accounting methods, principles, practices, or periods from those utilized in the preparation of the most recently filed Tax Returns or the Annual Financial Statements, except as required by GAAP, the statutory accounting principles and practices prescribed or permitted by the domiciliary state of the relevant Person, or applicable Tax law; and
- (j) The Company and the Subsidiaries shall not perform any act referenced by (or omit to perform any act which omission is referenced by) the terms of Section 3.8 hereof.
- (k) The Company shall provide Parent with reasonable assistance in investigating any legal obligations or risks relating to the Company's or any Subsidiary's membership in the GSM Association or the WIMAX Forum and in taking, in the sole discretion of Parent, actions to reduce the impact of such legal obligations or risks on the Parent or any subsidiary thereof (including, for example, Parent negotiating with either or both of the GSM Association and/or WIMAX Forum to exclude certain intellectual property rights of Parent or any subsidiary thereof from such legal obligations). Upon written request from Parent, if given no later than 21 calendar days after the date of signing of this Agreement, the Company shall cause the Company and all Subsidiaries (to the extent such Subsidiaries are members) to withdraw from either or both of the GSM Association and/or the WIMAX Forum.
- 5.2 <u>Pre-Closing Access</u>. Between the date hereof and the Closing Date, the Company and the Subsidiaries shall afford to Parent and its authorized representatives full access at all reasonable times and upon reasonable notice to the facilities, officers, employees, products, processes, technology, business and financial records, contracts, business plans, budget and projections, and other information regarding the Company and the Subsidiaries as Parent may reasonably request, and the work papers of Grant Thornton LLP, the Company's independent accountants, and otherwise provide such assistance as is reasonably requested by Parent in order that Parent may have a full opportunity to make such investigation and evaluation as it shall reasonably desire to make of the business and affairs of the Company and the Subsidiaries. In addition, the Company and its officers and directors shall cooperate fully (including providing introductions where necessary) with Parent to enable Parent to contact those third parties, including customers, prospective customers, specifying agencies, vendors or suppliers of the Company and the Subsidiaries, set forth on *Schedule 5.2* hereto.

Article VI

Additional Agreements

6.1 Company Stockholders' Written Consent.

- (a) The Company will take all action necessary under all applicable laws and the Company's Charter Documents and Governing Documents to solicit the stockholders holding the number of shares of each class and series of its Company Capital Stock entitled to be voted on the Merger necessary to achieve the Required Stockholder Vote to execute a written consent resolving to adopt this Agreement (the "Company Stockholders' Consent"). The Company will ensure that all actions taken to obtain the Company Stockholders' Consent will comply with all applicable laws. Any documents disseminated to the Company's stockholders in connection with the Company Stockholders' Consent, including any notice required to be delivered to stockholders who have not signed the Company Stockholders' Consent, shall be approved by Parent, which approval shall not be unreasonably withheld.
- (b) The notice accompanying the Company Stockholders' Consent or any proxy statement will include a statement to the effect that the board of directors of the Company recommends that the Company's stockholders adopt this Agreement (the recommendation of the Company's board of directors that the Company's stockholders vote to adopt this Agreement being referred to as the "Company Board Recommendation"). Subject to Section 6.5 hereof, the board of directors of the Company may not withdraw or modify the Company Board Recommendation in a manner adverse to Parent, and, no resolution by the board of directors of the Company or any committee thereof to withdraw or modify the Company Board Recommendation in a manner adverse to Parent may be adopted or proposed.
- 6.2 <u>Regulatory Filings</u>. As promptly as practicable after the execution of this Agreement, Parent and the Company shall make or cause to be made all filings and submissions under any laws, including the HSR Act and the other competition filings which apply to the transactions contemplated by this Agreement, applicable to such party for the consummation of the transactions contemplated herein. The Company will coordinate and cooperate with Parent, and Parent will coordinate and cooperate with the Company in exchanging such information necessary to make such filings. Any cooperation provided by any party to the other parties towards making the necessary filings and obtaining the necessary approvals shall not derogate from or release such party from its obligations to make the necessary filings and obtain the necessary approvals applicable to such party for the consummation of the transactions contemplated herein.
- 6.3 Conditions. Each of Parent and the Company shall take all commercially reasonable actions necessary or desirable to comply promptly with all legal requirements which may be imposed on such party with respect to the Merger and will promptly cooperate with and furnish such information to any other party hereto in connection with any such requirements imposed upon such other party in connection with the Merger. Each party will take all commercially reasonable actions to obtain (and will cooperate with the other parties in obtaining) any consent, authorization, order or approval of or any registration, declaration or filing with, or an exemption by any Governmental Entity, or other third party, required to be obtained or made by such party in connection with the Merger or the taking of any action contemplated thereby or by this Agreement, including all consents, waivers or approvals required under any Contracts disclosed in Section 3.13 of the Disclosure Schedule; *provided*, *however*, that Parent shall not be required to agree to any divestiture by Parent, the Company or any of Parent's subsidiaries, or affiliates, of shares of capital stock or of any business, assets or property of Parent or its subsidiaries or affiliates or of the Company or its affiliates or the imposition of any material limitation on the ability of any of them to conduct their business or to own or exercise control of such assets, properties and stock.

6.4 No Negotiations.

(a) Subject to Section 6.5 hereof, from and after the date of this Agreement until the earlier to occur of the Closing Date or termination of this Agreement pursuant to its terms, the Company shall not, and the Company will instruct its officers, directors, employees, agents, representatives and affiliates, not to, directly (i) solicit, initiate or knowingly encourage the submission of any Takeover Proposal from any Person (including any of its or their officers or employees, representatives, agents, or affiliates), other than the Parent or its affiliates or representatives, (ii) participate in any discussions or negotiations regarding, or furnish to any other Person (other than the Parent or its affiliates or representatives) any information with respect to, or otherwise cooperate in any way with, or assist or participate in, facilitate or knowingly encourage any effort or attempt by any other Person to do or seek any of the

foregoing or (iii) approve, endorse or recommend any Takeover Transaction (except to the extent specifically permitted pursuant to Section 6.5 hereof), or (iv) enter into any letter of intent or similar document or any contract, agreement or commitment contemplating or otherwise relating to any possible or proposed Takeover Proposal. The Company shall promptly notify Parent if any Takeover Proposal, or any inquiry from or contact with any Person with respect to a Takeover Proposal, is made and shall promptly provide Parent with material terms regarding the Takeover Proposal, including, but not limited to pricing and structure terms.

- (b) From and after the date of this Agreement until the earlier to occur of the Closing Date or termination of this Agreement pursuant to its terms, the Company shall suspend all activities related to the Company's proposed initial public offering.
 - 6.5 Fiduciary Obligations.
 - (a) The following terms will have the definitions set forth below:
 - (i) A "<u>Takeover Transaction</u>" shall mean any of the following transactions: (i) any transaction or series of related transactions with one or more third Persons involving: (A) any purchase from the Company or acquisition by any Person or "group" (as defined under Section 13(d) of the Securities and Exchange Act of 1934 and the rules and regulations thereunder) of more than 30% interest in the total outstanding voting securities of the Company or any tender offer or exchange offer that, if consummated, would result in any Person or group beneficially owning 30% or more of the total outstanding voting securities of the Company or any merger, consolidation or business combination involving the Company as a whole, or (B) any sale, lease (other than in the ordinary course of business consistent with past practice), exchange, transfer, license (other than in the ordinary course of business consistent with past practice), acquisition or disposition of more than 30% of the assets of the Company (including equity securities in any Subsidiary of the Company) on a consolidated basis, or (ii) any recapitalization, liquidation or dissolution of the Company consisting of more than 30% of the consolidated assets of the Company on a consolidated basis.
 - (ii) A "Takeover Proposal" shall mean any offer or proposal relating to a Takeover Transaction;
 - (iii) A "Superior Proposal" means an unsolicited, bona fide, written Takeover Proposal made by a third Person to acquire, directly or indirectly, pursuant to a tender offer, exchange offer, merger, consolidation or other business combination, (A) 50% or more of the assets of the Company on a consolidated basis or (B) 50% or more of the outstanding voting securities of the Company, and as a result of which, the stockholders of the Company immediately preceding such transaction would hold less than 50% of the aggregate equity interests in the surviving or resulting entity of such transaction (or its ultimate parent), which the Board of Directors of the Company has in good faith determined to be more favorable to the Company's stockholders (in their capacity as stockholders) than the terms of this Agreement (taking into account, among other things (1) the advice of its outside legal counsel and its financial adviser, and (2) all terms of such Takeover Proposal and this Agreement (as it may be proposed to be amended by Parent)) and to be reasonably capable of being consummated on the terms proposed, taking into account legal, financial, regulatory and other aspects of such Takeover Proposal and the Person making such Takeover Proposal including, if such Takeover Proposal involves any financing, the likelihood of obtaining such financing and the terms on which such financing may be secured.
- (b) Notwithstanding anything to the contrary contained in Section 6.4 hereof, in the event that the Company receives, prior to the Required Stockholder Vote (which may be obtained pursuant to the Company Stockholders' Consent), a bona fide Takeover Proposal which is determined by its board of directors to be, or is reasonably likely to lead to, a Superior Proposal, it may then take the following actions (but only (1) if and to the extent that (x) its board of directors concludes in good faith, after receipt of advice of its outside legal counsel, that the failure to do so would create a substantial risk of liability for breach of its fiduciary duties to the Company's stockholders under applicable law and (y) the Company has given Parent at least three business days' prior written notice of its intention to take any

of the following actions and of the identity of the Person or group making such Superior Proposal and the material terms and conditions of such Superior Proposal and (2) if it shall not have breached in any material respect any of the provisions of this Section 6.5 or Section 6.4 hereof):

- (i) furnish nonpublic information to the Person or group making such Superior Proposal, provided that (A) prior to furnishing any such nonpublic information, it receives from such Person or group an executed confidentiality agreement containing terms at least as restrictive as the terms contained in the Confidentiality Agreement, effective as of May 15, 2007, between Parent and the Company, and (B) contemporaneously with furnishing any such nonpublic information to such Person furnishes such nonpublic information to Parent (to the extent such nonpublic information has not been previously so furnished to Parent); and
- (ii) engage in negotiations with such Person or group with respect to such Superior Proposal; *provided*, *however*, in no event shall such party enter into any definitive agreement to effect such Superior Proposal.
- (c) In response to the receipt of a bona fide Takeover Proposal which is determined by the Company's board of directors to be a Superior Proposal, the board of directors may withhold, withdraw, amend or modify its recommendation in favor of approval and adoption of this Agreement and the Merger, and, in the case of a Superior Proposal that is a tender or exchange offer, may recommend that its stockholders accept such tender or exchange offer (any of the forgoing actions, whether by a board of directors or a committee thereof, a "*Change of Recommendation*") if the board of directors of the Company has concluded in good faith, after receipt of the advice of its outside legal counsel, that, in light of such Superior Proposal, the failure of the board of directors to effect a Change of Recommendation would create a substantial risk of liability for breach of its fiduciary duties to the Company's stockholders under applicable law.
- (d) Prior to announcing any Change of Recommendation pursuant to Section 6.5(c) hereof, the Company shall (A) provide to Parent three business days' prior written notice which shall (x) state expressly that it intends to effect a Change of Recommendation and (y) describe any material modifications to the material terms and conditions of the Superior Proposal and the identity of the Person or group making the Superior Proposal from the description of such terms and conditions and such Person contained in the notice required under Section 6.4 hereof; (B) make available to Parent all materials and information made available to the Person or group making the Superior Proposal in connection with such Superior Proposal, and (C) during the three business-day period commencing upon receipt of the notice described in this Section 6.5(d)(A), if requested by Parent, engage in good faith negotiations with Parent to amend this Agreement in such a manner that the Takeover Proposal which was determined to be a Superior Proposal no longer is a Superior Proposal.
- (e) If the board of directors of the Company effects a Change of Recommendation, the Company shall promptly notify Parent in writing of such Change of Recommendation. If the Company or Parent thereafter terminates this Agreement in accordance with Section 8.1(e) hereof, then the Company shall promptly pay the Termination Fee to Parent pursuant to Section 8.3 hereof.
- (f) Nothing contained in Section 6.4 hereof or elsewhere in this Agreement shall prohibit the Company from making any disclosure to the Company's stockholders or any public announcement if, in the good faith judgment of the board of directors of the Company, after consultation with outside counsel, failure to so disclose would violate any applicable law or constitute the breach of a fiduciary duty of the board of directors; provided that the Company shall not, except in accordance with the provisions of Section 6.5(c) hereof withdraw or modify, or propose to withdraw or modify, its recommendation of the Merger, or approve or recommend, or propose to approve or recommend, a Takeover Proposal.
- 6.6 <u>Notification of Certain Matters</u>. The Company or Parent, as the case may be, shall promptly notify the other party of (a) its obtaining of knowledge as to the matters set forth in clauses (i), (ii) and (iii) below, or (b) the occurrence, or failure to occur, of any event, which occurrence or failure to occur would be likely to cause (i) any

representation or warranty contained in this Agreement to be untrue or inaccurate in any material respect at any time from the date hereof to the Closing Date, (ii) any material failure of the Company or Parent, as the case may be, or of any officer, director, employee or agent thereof, to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it under this Agreement or (iii) the institution of any claim, suit, action or proceeding arising out of or related to the Merger or the transactions contemplated hereby; *provided*, *however*, that no such notification shall affect the representations or warranties of the parties or the conditions to the obligations of the parties hereunder.

- 6.7 Employee Benefits.
- (a) The following terms used in this Section 6.7 shall have the following meanings:
- (i) "<u>Change of Control Agreement</u>" means an agreement to provide change in control benefits which is specifically identified as a "Change of Control Agreement" in Section 3.13(a)(i) of the Disclosure Schedule, as such agreement is in effect on the date hereof.
- (ii) "<u>Continuing Bonus Plan</u>" means a Plan to provide bonus or incentive compensation which is specifically identified as a "Continuing Bonus Plan" in Section 3.18(a) of the Disclosure Schedule, as such Plan is in effect on the date hereof.
- (iii) "<u>Continuing Employee</u>" means each employee, whether active or on leave of absence, of the Company or one of its Subsidiaries immediately prior to the Effective Time who continues employment with Parent, the Surviving Corporation or one of their respective subsidiaries following the Effective Time.
- (iv) "<u>Continuing Health/Welfare Plan</u>" means a Plan to provide medical, surgical, health care, hospitalization, dental, vision, workers' compensation, life insurance, death, disability, legal services, sickness, accident, cafeteria plan or paid time off benefits which is specifically identified as a "Continuing Health/Welfare Plan" in Section 3.18(a) of the Disclosure Schedule, as such Plan is in effect on the date hereof.
- (v) "<u>Continuing Severance Program</u>" means a Plan to provide salary continuation or termination pay which is specifically identified as a "Continuing Severance Program" in Section 3.18(a) of the Disclosure Schedule, as such Plan is in effect on the date hereof.
- (b) During the period beginning at the Effective Time and ending on the one-year anniversary of the Effective Date, except as otherwise provided in any employment agreement with the Company or any Subsidiary, (i) Parent will, or will cause the Surviving Corporation to provide each Continuing Employee benefits pursuant to any Continuing Severance Program in accordance with the terms of such Continuing Severance Program, (ii) Parent will not, and will cause the Surviving Corporation not to, require any Continuing Employee to relocate his or her workplace to a workplace more than 50 miles from the location of such Continuing Employee's workplace as of the Effective Time, and (iii) Parent will, or will cause the Surviving Corporation to provide each Continuing Employee with benefits pursuant to any Change of Control Agreement in accordance with the terms of such Change of Control Agreement.
- (c) During the period beginning at the Effective Time and ending on December 31, 2007, except as otherwise provided in any employment agreement with the Company or any Subsidiary, Parent will, or will cause the Surviving Corporation to provide each Continuing Employee, as long as such Continuing Employee remains employed by Parent, the Surviving Corporation or any of their respective subsidiaries, benefits pursuant to (i) any Continuing Bonus Plan in accordance with the terms of such Continuing Bonus Plan and (ii) any Continuing Health/Welfare Plan in accordance with the terms of such Continuing Health/Welfare Plan, (iii) the Company's 401(k) plan (to the extent that, and so long as, such Continuing Employee is enrolled in the Company's 401(k) plan). Bonuses earned as of December 31, 2007 under a Continuing Bonus Plan will be paid in accordance with such Continuing Bonus Plan even though payment of the bonus may occur after December 31, 2007. Parent agrees that it will not take any action specifically designed to impair the ability of Continuing Employees to earn the bonuses provided for in the Continuing

Bonus Plans for the primary purpose of minimizing the payments under such Continuing Bonus Plans. The Company, however, acknowledges that the transactions contemplated in this Agreement constitute a transfer of control to Parent and that, following the Closing Date, Parent and its representatives will exercise their business judgment regarding the operations of the Surviving Corporation in a manner consistent with Parent's duties to its shareholders.

- (d) With respect to employee benefit plans of Parent in which Continuing Employees become eligible to participate following the Effective Time (each, a "Parent Plan"), Parent will, or will cause the Surviving Corporation to, (i) with respect to each Parent Plan that is a medical or health plan, waive any exclusions for pre-existing conditions under such Parent Plan that would result in a lack of coverage for any condition for which the applicable Continuing Employee would have been entitled to coverage under the corresponding Plan in which such Continuing Employee was an active participant immediately prior to his or her transfer to Parent Plan and (ii) recognize service of the Continuing Employee with the Company for purposes of eligibility to participate and vesting credit, to the extent that such service was recognized for that purpose under the analogous Plan prior to such transfer; provided, however, that the foregoing shall not apply to the extent it would result in duplication of benefits. Parent will, or will cause the Surviving Corporation to, credit Continuing Employees who have between four and five years of service with the Company as of the Effective Date with additional service such that the Continuing Employee is treated as having five years of service for purposes of Parent's paid time off policy; provided that the amount of additional service credited pursuant to this sentence shall not be taken into account for purposes of any final paid time off payment in the event the Continuing Employee terminates employment with the Company before the one year anniversary of the Effective Date.
- (e) Parent will, and will cause the Surviving Corporation to, honor the annual compensation increases going into effect on January 1, 2008 determined by the Company as set forth in Section 3.8(j) of the Disclosure Schedule.
- (f) Parent will, and will cause the Surviving Corporation to, continue to provide payroll services (including payment and reporting of all applicable federal, state and local withholding or employment taxes, disbursements of net payroll to all Continuing Employees, maintenance of any direct deposit agreements as in effect as of the Effective Time, transfer to the Company's 401(k) trust the amounts withheld pursuant to the Company's 401(k) plan, and implementation of elections made under the Continuing Health/Welfare Plans) with respect to Continuing Employees for the period beginning on the Effective Date and ending on December 31, 2007.
- (g) Nothing in this Agreement will be construed (i) to create a right in any Continuing Employee to employment with Parent, the Surviving Corporation or any other subsidiary of Parent or the Surviving Corporation and (ii) as an amendment or modification to any employment agreement in effect as of the Effective Time. The employment of Continuing Employees will be "at will" employment to the extent employment is "at will" as of the Effective Time. Nothing in this Agreement will be construed so as to vest in any Continuing Employee any third-party beneficiary right in the agreements of the parties hereto contained in this Section 6.7 or otherwise.

6.8 Indemnification of Officers and Directors.

(a) From and after the Effective Time, Parent shall cause the Surviving Corporation to fulfill and honor in all respects the obligations of the Company pursuant to any indemnification provisions under the certificate of incorporation and bylaws of the Company as in effect on the date of this Agreement and pursuant to any indemnity agreements between the Company and such person as in effect on the date of this Agreement (the persons entitled to be indemnified pursuant to such provisions, and all other current and former directors and officers of the Company, being referred to collectively as the "D&O Indemnified Parties"). From and after the Effective Time through the sixth anniversary of the of the date the Effective Time occurs, Parent shall cause the certificate of incorporation and bylaws of Merger Sub and the Surviving Corporation to contain the provisions with respect to indemnification and exculpation from liability set forth in the Company's certificate of incorporation and bylaws on the date of this Agreement, which provisions shall not be amended, repealed or otherwise modified during such period in any manner that would adversely affect the rights thereunder of any D&O Indemnified Party. Parent shall honor in accordance

with their terms all indemnification agreements entered into by the Company with any current or former director, officer or agent that are in effect prior to the execution of this Agreement.

- (b) Parent and the Surviving Corporation jointly and severally agree to pay all expenses, including attorneys' fees, that may be incurred by the D&O Indemnified Parties in enforcing the indemnity and other obligations provided for in this Section 6.8, provided that such D&O Indemnified Parties are entitled to the indemnities sought. For six years after the Effective Time, Parent shall maintain in effect the current level and scope of directors' and officers' liability insurance or a tail insurance policy of the same level or scope for the six year period, in each case covering those persons who are covered by the Company's directors' and officers' liability insurance policy as of the Effective Time (a copy of which has been heretofore delivered to Parent); *provided*, *however*, that in no event shall Parent be required to expend in any one year an amount in excess of 200% of the annual premium currently paid by the Company for such insurance, and *provided*, *further*, that if the annual premiums of such insurance coverage exceed such amount, Parent shall be obligated to obtain a policy with the greatest coverage available for a cost not exceeding such amount. This Section 6.8 shall survive the consummation of the Merger and the Effective Time, is intended to benefit and may be enforced by the Company, Parent the Surviving Corporation and the D&O Indemnified Parties, and shall be binding on all successors and assigns of Parent and the Surviving Corporation.
- 6.9 <u>Code Section 409A Compliance</u>. Before Closing, the Company shall use reasonable efforts to amend and otherwise bring into documentary compliance with Code Section 409A all of the Company's Plans and arrangements (including, but not limited to, employment agreements and change of control agreements) that are subject to Code Section 409A.
- 6.10 Form S-8. Within 30 days after the Effective Time, Parent shall file a registration statement on Form S-8 (or any successor or other appropriate form) which will register the shares of common stock of Parent subject to the Parent Rollover Options to the extent permitted by federal securities laws and shall use its reasonable best efforts to maintain the effectiveness of such registration statement or registration statements (and maintain the current status of the prospectus or prospectuses contained therein) for so long as such options remain outstanding, subject in each case to policies and practices generally applicable to options to purchase shares of common stock of Parent at such time.

Article VII

Conditions to Closing

- 7.1 <u>Conditions to Obligations of Each Party to Effect the Merger</u>. The respective obligations of each party to this Agreement to effect the Merger shall be subject to the satisfaction of each of the following conditions on or before the Closing Date:
- (a) <u>No Injunctions or Restraints</u>. No judgment, order, decree or law entered, enacted, promulgated, enforced or issued by any court or other Governmental Entity of competent jurisdiction or other legal restraint or prohibition shall be in effect (i) imposing or seeking to impose material sanctions, damages, or liabilities directly arising out of the Merger on Parent or the Company or any of their officers or directors; or (ii) preventing the consummation of the Merger.
- (b) <u>Governmental Action</u>. No action or proceeding shall be threatened, instituted or pending by any Governmental Entity challenging or seeking to prevent or delay consummation of or seeking to render unenforceable the Merger, asserting the illegality of the Merger or any material provision of this Agreement or seeking material damages in connection with the transactions contemplated hereby which continues to be outstanding.
- (c) <u>Governmental Approvals</u>. The applicable waiting period under the HSR Act shall have expired or been terminated, and all other material governmental filings, authorizations, consents, orders or approvals that are required for the consummation of the transactions contemplated hereby will have been duly filed and, as required, obtained.

- 7.2 <u>Additional Conditions to Parent's Obligations</u>. The obligation of Parent to consummate the transactions contemplated by this Agreement is subject to the satisfaction or waiver (in Parent's sole discretion) of each of the following conditions on or before the Closing Date:
- (a) Representations and Warranties True and Correct. The representations and warranties set forth in Section 3.3 hereof shall be true and correct at and as of the Closing Date as though then made and as though the Closing Date had been substituted for the date of this Agreement. The representations and warranties set forth in Article III hereof (other than the representations and warranties set forth in Section 3.3 hereof) shall be true and correct in all material respects at and as of the Closing Date as though then made and as though the Closing Date had been substituted for the date of this Agreement throughout such representations and warranties except for inaccuracies of such representations, the circumstances giving rise to which, individually or in the aggregate, do not constitute and could not reasonably be expected to have a Material Adverse Effect (without giving effect to any limitation as to "materiality" or "Material Adverse Effect" set forth therein and without taking into account any disclosures by the Company of discoveries, events or occurrences arising on or after the date hereof), except that any such representation or warranty made as of a specified date (other than the date hereof) shall only need to have been true on and as of such date;
- (b) <u>Covenants Performed</u>. The Company shall have performed in all material respects all of the covenants and agreements required to be performed and complied with by it under this Agreement prior to the Closing;
- (c) No Injunctions or Restraints. No judgment, order, decree or law entered, enacted, promulgated, enforced or issued by any court or other Governmental Entity of competent jurisdiction or other legal restraint or prohibition shall be in effect (i) imposing or seeking to impose material limitations on the ability of Parent to acquire or hold or to exercise full rights of ownership of any securities of the Company; (ii) imposing or seeking to impose material limitations on the ability of Parent or its affiliates to combine and operate the business and assets of the Company; (iii) imposing or seeking to impose other material sanctions, damages, or liabilities directly arising out of the Merger on Parent or any of its officers or directors; or (iv) requiring or seeking to require divestiture by Parent of any portion of the business, assets or property of the Company or any Subsidiary or of Parent;
 - (d) Stockholder Approval. This Agreement shall have been duly adopted by the Required Stockholder Vote;
- (e) <u>Consents Obtained</u>. The Company shall have obtained, or caused to be obtained the consents and approvals identified in *Schedule 7.2(e)* hereto;
- (f) <u>Consents of Holders of Company Options</u>. The Company shall have obtained, or caused to be obtained, the executed consent and approval to the treatment of such Company Options contemplated by Sections 2.4(a)-2.4(c) hereof, as applicable, substantially in the form set forth in <u>Exhibit D-1</u> or <u>Exhibit D-2</u> hereto, as applicable, of holders of Company Options representing 95% of the Company Common Stock underlying outstanding Company Options and such executed consents and approvals shall be in full force and effect;
- (g) <u>Consent of Holders of Qualified Warrants</u>. The Qualified Warrants shall have been amended to allow for the treatment contemplated by Section 2.4(d) of this Agreement;
- (h) <u>Conversion of Convertible Note</u>. The Company shall have effected the conversion of the Convertible Notes, in the aggregate original principal amount of \$6,930,000, held by Alvarion Inc. (the "<u>Convertible Notes</u>") into shares of Company Common Stock, pursuant to the terms of the Convertible Notes;
- (i) <u>No Material Adverse Change</u>. Between the date of this Agreement and the Closing Date, there shall not have occurred any event or circumstance (without taking into account any disclosures by the Company of discoveries, events or occurrences arising on or after the date hereof), which would result in or would be reasonably likely to have a Material Adverse Effect;

- (j) <u>No Damage to Properties</u>. There shall have been no damage, destruction or loss of or to any property or properties owned or used by the Company or any Subsidiary, whether or not covered by insurance, which, in the aggregate, has, or would be reasonably likely to have, a Material Adverse Effect;
- (k) <u>Key Employee Offer Letters</u>. The offer letters executed by the employees of the Company identified in *Schedule 7.2(k)* hereto (the "<u>Key Employees</u>") and delivered to Parent on the date hereof shall be in full force and effect, and none of such Key Employees shall have indicated any intention of not fulfilling his obligations thereunder;
- (l) <u>Appraisal Rights</u>. The Company shall have mailed the notice required under Section 262(d)(2) of the DGCL and at least 20 days shall have elapsed from the date of mailing of such notice. As of the Closing Date, holders of no more than 5% of the Company Capital Stock (calculated in the manner set forth in *Schedule 7.2(l)* hereof) outstanding immediately prior to the Closing shall have exercised any appraisal rights under the DGCL;
- (m) <u>Withdrawal from Trade Groups</u>. None of the Company and the Subsidiaries shall be a member of any trade group with respect to which the Parent has provided a written withdrawal request pursuant to Section 5.1(k); and
- (n) <u>Delivery of Certain Documents</u>. On the Closing Date, the Company shall have delivered to Parent all of the following:
 - (i) a certificate of the Chief Executive Officer and the Chief Financial Officer of the Company, dated the Closing Date, stating that the conditions precedent set forth in Sections 7.2(a) and 7.2(b) above have been satisfied:
 - (ii) a copy of (A) the text of the resolutions adopted by the board of directors of the Company authorizing the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated by this Agreement, including the Merger, (B) the Charter Documents and Governing Documents of the Company and each of the Subsidiaries, along with a certificate executed by the Secretary of the Company, certifying to Parent that such copies are true, correct and complete copies of such resolutions, Charter Documents and Governing Documents, respectively, and that such resolutions, Charter Documents and Governing Documents were duly adopted and have not been amended or rescinded and (C) an incumbency certificate executed on behalf of the Company by the Secretary of the Company certifying the signature and office of each officer executing this Agreement and such other agreements executed on behalf of the Company contemplated in this Agreement;
 - (iii) copies of the third party and governmental consents and approvals referred to in Sections 7.1(c) and 7.2(e) above;
 - (iv) the Company's and each Subsidiary's minute books, stock transfer records, corporate seal and other materials related to the Company's and each Subsidiary's corporate administration;
 - (v) a copy, fully executed by Stockholders' Representative, of the Escrow Agreement;
 - (vi) Parent shall have received the resignation in writing of the directors and officers of the Company and the Subsidiaries effective as of the Closing Date and each director shall have waived any and all claims against the Company and the Subsidiaries, except claims for indemnification pursuant to Section 6.8 hereof and customary expenses for board service, as of the Closing Date;
 - (vii) a certificate pursuant to Treasury Regulations Section 1.897-2(h) or certificates pursuant to Treasury Regulations Section 1.1445-2(b)(2), reasonably acceptable to Parent and duly executed by the relevant Persons;
 - (viii) copies of the consents described in Section 7.2(f) hereof and the amendment described in Section 7.2(g); and

- (ix) copies of the Company Stockholders' Consent representing the Required Stockholder Vote.
- 7.3 <u>Additional Conditions to the Company's Obligations</u>. The obligations of the Company to consummate the transactions contemplated by this Agreement are subject to the satisfaction or waiver (in the Company's sole discretion) of each of the following conditions on or before the Closing Date:
- (a) Representations and Warranties True and Correct. The representations and warranties set forth in Article IV hereof will be true and correct in all material respects at and as of the Closing as though then made and as though the Closing Date had been substituted for the date of this Agreement throughout such representations and warranties except for inaccuracies of such representations, the circumstances giving rise to which, individually or in the aggregate, do not constitute and could not reasonably be expected to have a Material Adverse Effect (without giving effect to any limitation as to "materiality" or "material adverse effect" set forth therein and without taking into account any disclosures by the Parent of discoveries, events or occurrences arising on or after the date hereof), except that any such representation or warranty made as of a specified date (other than the date hereof) shall only need to have been true on and as of such date:
- (b) <u>Covenants Performed</u>. Parent shall have performed in all material respects all the covenants and agreements required to be performed by it under this Agreement prior to the Closing;
- (c) <u>No Material Adverse Change</u>. Between the date of this Agreement and the Closing Date, there shall not have occurred any event or circumstance, which would result in or would be reasonably likely to have a Material Adverse Effect on Parent or Merger Sub (for purposes of this Section 7.3(c) and Section 7.3(a) only, references made to the Company shall be replaced by references to Parent in the definition of Material Adverse Effect in Section 3.1(a) herein); and
 - (d) <u>Delivery of Certain Documents</u>. On the Closing Date, Parent will have delivered to the Company:
 - (i) a certificate of an officer of Parent dated the Closing Date, stating that the conditions precedent set forth in Sections 7.3(a) and 7.3(b) above have been satisfied;
 - (ii) a copy, fully executed by Parent and the Escrow Agent, of the Escrow Agreement; and
 - (iii) copies of the governmental consents and approvals referred to in Sections 7.1(c) above.

Article VIII

Termination

- 8.1 Termination. This Agreement may be terminated at any time prior to the Closing:
- (a) by the mutual consent of Parent and the Company;
- (b) by either Parent or the Company if there has been a material misrepresentation, material breach of warranty or breach of covenant on the part of the Company, on the one hand, or Parent, on the other, in the representations, warranties and covenants set forth in this Agreement and such material misrepresentation or material breach of warranty or covenant is not cured by the breaching party within 30 days following the receipt by the breaching party of a notice from the non-breaching party of any such breach;
- (c) by either Parent or the Company if the transactions contemplated hereby have not been consummated by January 31, 2008; *provided*, *that*, neither Parent nor the Company will be entitled to terminate this Agreement pursuant to this Section 8.1(c) if Parent's or the Company's, respectively, willful breach of this Agreement has prevented the consummation of the transactions contemplated hereby;

- (d) by Parent if, after the date hereof, there shall have been a Material Adverse Effect or if after the date hereof, an event shall have occurred which, so far as reasonably can be foreseen, would result in any Material Adverse Effect; except to the extent that the Company is capable of taking action to cure the Material Adverse Effect and the Company has taken such action within 30 days following written notice from Parent that such a change or such an event has occurred; or
- (e) by the Company or Parent if the board of directors of the Company effects a Change of Recommendation in response to a Takeover Proposal but only after the Company provides written notice to Parent advising Parent that the board of directors of the Company has received a Takeover Proposal and specifying the material terms and conditions of such Takeover Proposal and identifying the Person making such Takeover Proposal.
- 8.2 Effect of Termination. In the event of termination of this Agreement by either Parent or the Company as provided in Section 8.1 hereof, all provisions of this Agreement shall terminate and there shall be no liability on the part of Parent or the Company or their respective shareholders, stockholders, officers, or directors, *except that*: (i) Sections 8.3 (termination fee), 10.1 (press releases), 10.2 (expenses) and 10.12 (governing law) hereof shall survive indefinitely, and (ii) notwithstanding the foregoing, the parties shall remain liable for their willful or fraudulent breaches of this Agreement prior to the time of such termination.
- 8.3 <u>Termination Fee</u>. In the event that the Company or Parent terminates this Agreement as provided in Section 8.1(e) hereof, the Company shall pay, within three business days of such termination, the sum of Five Million Dollars (\$5,000,000) in same day, readily available funds (the "<u>Termination Fee</u>") to Parent.

Article IX

Survival; Indemnification

9.1 Survival of Representations and Warranties. Notwithstanding any investigation made by or on behalf of any of the parties hereto or the results of any such investigation and notwithstanding the participation of such party in the Closing, the representations and warranties contained in Article I, Article II, Article III and Article IV hereof shall, except as provided below, survive the Closing for a period of 15 months from the Closing Date; provided, however, that if a party shall have made a claim for indemnification with respect to any specific representation(s) or warranty(ies) hereunder prior to the 15-month anniversary of the Closing Date and such claim has not been completely and finally resolved prior to such anniversary, then such claim shall survive and continue to be fully actionable until such time as it is completely and finally resolved as if the specific representation(s) or warranty(ies) upon which it is based had not vet expired. Notwithstanding the preceding sentence, the representations and warranties set forth in Sections 3.3, 3.12 and 3.14 shall survive the Closing for a period of three years from the Closing Date, *provided*, however, that, with respect to any specific representation or warranty set forth in Sections 3.3, 3.12 or 3.14, if a party shall have made a claim for indemnification with respect to any specific representation(s) or warranty(ies) hereunder prior to the three-year anniversary of the Closing Date and such claim has not been completely and finally resolved prior to such anniversary, then such claim shall survive and continue to be fully actionable until such time as it is completely and finally resolved as if the specific representation(s) or warranty(ies) upon which it is based had not yet expired. This Section 9.1 shall not limit any covenant or agreement of the parties that by its terms contemplates performance after Closing.

9.2 Indemnification by Stockholders.

(a) Subject to the limitations of Section 9.2(b) hereof, holders of Company Capital Stock and Qualified Warrants immediately prior to the Effective Time shall indemnify in full Parent, the Company and their respective officers, directors, employees, agents, shareholders, stockholders and subsidiaries (collectively, the "*Parent Indemnified Parties*") and hold them harmless from and against any claim, demand, loss, liability, obligation, deficiency, action, damage, expense or cost (including reasonable legal expenses), whether or not actually incurred or paid (collectively "*Losses*"), which the Parent Indemnified Parties suffer, sustain or become subject to as a result of (i) any inaccuracy or

misrepresentation in or breach of any of the representations and warranties of the Company contained in this Agreement or in the Disclosure Schedule or closing certificates delivered or to be delivered by or on behalf of the Company pursuant to the terms of this Agreement (collectively, the "Related Documents"), (ii) any breach of, or failure to perform, any covenant of the Company contained in this Agreement or any of the Related Documents. (iii) all Taxes of the Company and the Tax Affiliates for all taxable periods ending on or prior to the Closing Date and the portion through the end of the Closing Date (determined with respect to periodic Taxes not based on income or receipts (e.g. property taxes) based on the proportional number of days in the relevant period and with respect to income Taxes based on a closing of the books) for any taxable period that includes (but does not end on) the Closing Date. (iv) any Claims (as defined in Section 9.5(a) hereof) or threatened Claims against any Parent Indemnified Party arising out of the actions or inactions of the Company with respect to the Company's business or the Real Property prior to the Closing Date. (v) the failure of the Company to obtain, prior to the Effective Time, the consent of any holder of a Company Option or a Company Warrant to the treatment of such Company Option or Company Warrant as contemplated by Section 2.4 of this Agreement, (vi) any matter described under the heading "Specific Patent Issues" in Section 3.14(b) of the Disclosure Schedule which is specifically identified as covered by this Section 9.2(a)(vi) or (vii) the matters described on Schedule 7.2(1) hereto (collectively, the "Parent Losses"), provided, however, that any indemnification for Parent Losses for matters described in Sections 9.2(a)(v)-(vii) hereof shall be subject to the same provisions and conditions of this Article IX as if such matters were brought under Section 9.2(a)(i) hereof; provided. further that Parent Losses shall not include any amount for "punitive", "consequential" or "special damages" or damages for "lost profits" except to the extent that (x) a Claim alleges such forms of damages. (v) a Parent Indemnified Party is deemed responsible therefor pursuant to such Claim and (z) the responsible Parent Indemnified Party incurs Losses as a result thereof. The calculation of any such Parent Loss will reflect (A) the amount of any Tax benefit actually recognized by the Parent Indemnified Parties for the United States federal and state income Tax purposes in the year in which such Parent Loss is suffered or incurred and which is determined by the applicably Parent Indemnified Party's Tax Return preparers to be without material risk of being disallowed upon audit and (B) the amount of any insurance proceeds received by the Parent Indemnified Party in respect of such Parent Loss. For purposes of (A) above, a Tax benefit is "actually realized" if it can be utilized to reduce such Parent Indemnified Party's Tax liability with respect to the Tax year in which the Parent loss is suffered or incurred, and, for the avoidance of doubt, the mere incurring or increase in a capital or net operating loss or similar Tax asset, without a resulting reduction in Tax in that year, will not be deemed a Tax benefit for purposes of this provisions. The indemnification provided by this Section 9.2(a) shall be satisfied solely from the Escrow Amount.

(b) Holders of Company Capital Stock and Qualified Warrants immediately prior to the Effective Time will be liable to the Parent Indemnified Parties for Parent Losses resulting from an inaccuracy or misrepresentation in or breach of any of the representations and warranties of the Company contained in this Agreement or in the Related Documents or from the matters described in Sections 9.2(a)(v)-(vii) hereof (i) only if Parent delivers to Stockholders' Representative a written notice, pursuant to Sections 9.4 and 9.5 hereof, as applicable, with respect to such Parent Indemnified Party's Claim to be indemnified for such Parent Losses prior to 15 months after the Closing Date (or. in the case of Parent Losses attributable to (A) an inaccuracy or misrepresentation in breach of the representations and warranties contained in Sections 3.3, 3.12 or 3.14 or (B) the matters described in Sections 9.2(a)(v)-(vii) hereof, prior to three years after the Closing Date), and (ii) only if the aggregate amount of all such Parent Losses exceeds \$775,000 (the "Basket Amount"), in which case such holders shall be obligated to indemnify Parent Indemnified Parties for the aggregate amount of all Parent Losses, provided, however, that no Parent Indemnified Party shall make any claim for Losses pursuant to Section 9.2(a) hereof which individually does not exceed \$15,000, and such claims not meeting this threshold shall not be applied in calculating the Basket Amount specified above, further provided, however, that the Basket Amount shall not apply to (X) breaches, inaccuracies or misrepresentations of representations and warranties set forth in any of Sections 3.3 and 3.12 hereof or (Y) the matters described in Sections 9.2(a)(v)-(vii) hereof. Except for fraud in connection with any breach or misrepresentation by the Company. (i) the aggregate liability of all holders of Company Capital Stock and Qualified Warrants immediately prior to the Effective Time for Parent Losses resulting from inaccuracies or misrepresentations in or breaches of any of the representations and warranties of the Company contained in this Agreement or from the matters described in Sections 9.2(a)(v)-(vii) hereof shall not exceed an amount equal to the Escrow Amount (the "Cap") and (ii) the aggregate liability of each holder of Company Capital

Stock and Qualified Warrants for such Parent Losses shall not exceed the amount contributed to the Escrow Amount by Parent on behalf of such holder; *provided, however*, that nothing in clause (ii) of this sentence shall be construed as limiting in any manner the availability of the Escrow Amount to satisfy the indemnification obligations of the holders of Company Capital Stock and Qualified Warrants to Parent under this Article IX.

9.3 Indemnification by Parent.

- (a) Subject to the limitations of Section 9.3(b) hereof, Parent agrees to indemnify in full holders of Company Capital Stock, Qualified Options and Qualified Warrants immediately prior to the Effective Time and their respective officers, directors, employees, agents and subsidiaries (collectively, the "Holder Indemnified Parties") and hold them harmless from and against any Losses which any of the Holder Indemnified Parties may suffer, sustain or become subject to, arising from or relating to (i) any inaccuracy or misrepresentation in or breach of any of the representations and warranties of Parent contained in this Agreement or other Related Documents or (ii) any breach of, or failure to perform, any covenant of Parent contained in this Agreement (collectively, the "Holder Losses") provided, however, that Holder Losses shall not include any amount for "punitive", "consequential" or "special damages" or damages for "lost profits" except to the extent that (x) a Claim alleges such forms of damages, (y) a Holder Indemnified Party is deemed responsible therefor pursuant to such Claim and (z) the responsible Holder Indemnified Party incurs Losses as a result thereof. The calculation of any such Holder Loss will reflect (A) the amount of any Tax benefit actually recognized by the Holder Indemnified Parties for the United States federal and state income Tax purposes in the year in which such Holder Loss is suffered or incurred and which is determined by the applicably Holder Indemnified Party's Tax Return preparers to be without material risk of being disallowed upon audit and (B) the amount of any insurance proceeds received by the Holder Indemnified Party in respect of such Holder Loss. For purposes of (A) above, a Tax benefit is "actually realized" if it can be utilized to reduce such Holder Indemnified Party's Tax liability with respect to the Tax year in which the Holder Loss is suffered or incurred, and, for the avoidance of doubt, the mere incurring or increase in a capital or net operating loss or similar Tax asset, without a resulting reduction in Tax in that year, will not be deemed a Tax benefit for purposes of this provisions.
- (b) Parent will be liable to the Holder Indemnified Parties for Holder Losses from an inaccuracy or misrepresentation in or breach of any of the representations and warranties of the Parent contained in this Agreement or in the Related Documents (i) only if Stockholders' Representative delivers to Parent a written notice, pursuant to Section 9.4 or 9.5 hereof, as applicable, with respect to such Holder Indemnified Party's claim to be indemnified for such Holder Losses prior to 15 months after the Closing Date, and (ii) only if the aggregate amount of all such Holder Losses exceeds the Basket Amount, in which case Parent shall be obligated to indemnify the Holder Indemnified Parties for the aggregate amount of all Holder Losses, *provided*, *however*, that no Holder Indemnified Party shall make any claim for Losses pursuant to Section 9.3(a) hereof which individually does not exceed \$15,000, and such claims not meeting this threshold shall not be applied in calculating the Basket Amount specified above. Except for fraud in connection with any breach or misrepresentation by Parent, Parent's aggregate liability for Holder Losses resulting from inaccuracies or misrepresentations in or breaches of any of the representations and warranties of Parent contained in this Agreement shall not exceed the Cap.
- 9.4 Notice of Claims. Any written notice required to be delivered to the applicable Indemnifying Party under Section 9.2 or 9.3 hereof, as the case may be, shall specify in reasonable detail (i) the representation and warranty or covenant that is alleged to have been inaccurate or to have been breached, (ii) the basis for such allegation and (iii) if known, the aggregate amount of the Losses for which a claim is being made under this Article IX or, to the extent that such Losses are not known or have not been incurred at the time such claim is made, an estimate, to be prepared in good faith, of the aggregate potential amount of such Losses. Written notice to such Indemnifying Party of the existence of a claim shall be given by the Indemnified Party promptly after the Indemnified Party first receives notice of the potential claim; *provided*, *however*, that the Indemnified Party shall not be foreclosed from seeking indemnification pursuant to this Article IX by any failure to provide such prompt notice of the existence of a claim to the applicable Indemnifying Party or to supply such relevant supporting documentation, except and only to the extent that such Indemnifying Party has been materially damaged or prejudiced as a result of such delay.

- 9.5 <u>Method of Asserting Claims</u>. As used herein, an "<u>Indemnified Party</u>" shall refer to a Parent Indemnified Party or Holder Indemnified Party, as applicable, and the "<u>Indemnifying Party</u>" shall refer to the party or parties hereto obligated to indemnify such Indemnified Party.
- (a) In the event that any Indemnified Party is made a defendant in or party to any action or proceeding, judicial or administrative, instituted by any third party for the liability or the costs or expenses of which are Losses (any such third party action or proceeding being referred to as a "Claim"), the Indemnified Party shall give the Indemnifying Party prompt notice thereof. The failure to give such notice shall not affect any Indemnified Party's ability to seek reimbursement unless such failure has materially damaged or prejudiced the Indemnifying Party's ability to defend successfully a Claim. The Indemnifying Party shall be entitled to contest and defend such Claim: provided, that the Indemnifying Party (i) has a reasonable basis for concluding that such defense may be successful and (ii) can diligently contest and defend such Claim. Notice of the intention to so contest and defend shall be given by the Indemnifying Party to the Indemnified Party within 20 business days after the Indemnified Party's notice of such Claim (but, in all events, at least five business days prior to the date that an answer to such Claim is due to be filed). Such contest and defense shall be conducted by reputable attorneys retained by the Indemnifying Party. The Indemnified Party shall be entitled at any time, at its own cost and expense (which expense shall not constitute a Loss unless the Indemnified Party reasonably determines that the Indemnifying Party is not adequately representing or. because of a conflict of interest, may not adequately represent, any interests of any Indemnified Party, and only to the extent that such expenses are reasonable), to participate in such contest and defense and to be represented by reputable attorneys of its or their own choosing. If the Indemnified Party elects to participate in such defense, the Indemnified Party shall cooperate with the Indemnifying Party to the extent reasonably requested by the Indemnifying Party in the contest and defense of such Claim, including providing reasonable access (upon reasonable notice) to the books, records and employees of the Indemnified Party if relevant to the defense of such Claim; provided that, such cooperation shall not unduly disrupt the operations of the business of the Indemnified Party or cause the Indemnified Party to waive any statutory or common law privileges, breach any confidentiality obligations owed to third parties or otherwise cause any trade secret or confidential information of such Indemnified Party to become public. Neither the Indemnified Party nor the Indemnifying Party may concede, settle or compromise any Claim without the consent of the other party, which consent will not be unreasonably withheld. Notwithstanding the foregoing, (i) if a Claim seeks the issue of an injunction, the specific election of an obligation or similar remedy, or (ii) if the subject matter of a Claim relates to the ongoing business of any of the Indemnified Parties, which Claim, if decided against any of the Indemnified Parties, would materially adversely affect the ongoing business or reputation of any of the Indemnified Parties, then, in each such case, the Indemnified Parties alone shall be entitled to contest, defend and settle such Claim in the first instance and, if the Indemnified Parties do not contest, defend or settle such Claim, the Indemnifying Party shall then have the right to contest and defend (but not settle) such Claim.
- (b) In the event any Indemnified Party should have a claim against the Indemnifying Party that does not involve a Claim, the Indemnified Party shall deliver a notice of such claim with reasonable promptness to the Indemnifying Party or, if holders of the Company Capital Stock and Qualified Warrants immediately prior to the Effective Time are the Indemnifying Party, to the Stockholders' Representative. If the Indemnifying Party notifies the Indemnified Party that the Indemnifying Party does not dispute the claim described in such notice or fails to notify the Indemnified Party in writing within 20 business days after delivery of such notice by the Indemnified Party whether the Indemnifying Party disputes the claim described in such notice, the Loss in the amount specified in the Indemnified Party's notice will be conclusively deemed a liability of the Indemnifying Party, and the Indemnifying Party shall pay (subject to the Basket Amount, to the extent applicable), the amount of such Loss to the Indemnified Party on demand. If the Indemnifying Party has timely disputed its liability with respect to such claim, Stockholders' Representative and an authorized officer of Parent will proceed in good faith to negotiate a resolution of such dispute, and if not resolved through the negotiations of such individuals within 20 business days after the delivery of the Indemnifying Party's notice that it disputes such claim, such dispute shall be resolved fully and finally in Minneapolis, Minnesota, by an arbitrator selected pursuant to and an arbitration governed by Commercial Arbitration Rules of the American Arbitration Association, as modified herein. The parties will jointly appoint a mutually acceptable independent arbitrator, seeking assistance in such regard from the American Arbitration Association. The arbitrator shall resolve

the dispute based on the governing law, without regard to its rules of evidence, and judgment upon the award rendered by such arbitrator may be entered in any court of competent jurisdiction. Each of Parent, on the one hand, and Stockholders' Representative, on the other, shall bear its own fees and expenses in connection with such arbitration and shall bear 50% of the fees and expenses of the arbitrator. Stockholders' Representative shall be entitled to deduct the amount of its fees and expenses incurred in connection with pursuing or defending Claims pursuant to this Article IX from the income received in respect of the Escrow Fund.

9.6 Sole and Exclusive Remedy. After the Closing, the rights set forth in this Article IX shall be each party's sole and exclusive remedies against the other parties hereto for misrepresentations, inaccuracies or breaches of representations and warranties contained in this Agreement and the Related Documents or for the matters described in Sections 9.2(a)(v)-(vii) hereof. Notwithstanding the foregoing, nothing herein shall prevent any of the parties hereto from bringing an action based upon allegations of fraud in connection with this Agreement and the Related Documents. In the event such action is brought, Parent, on the one hand, and Stockholders' Representative, on the other, shall each bear its own fees and expenses in connection with such action.

9.7 Stockholders' Representative.

- (a) Robert R. Ackerman, Jr. (the "Stockholders' Representative") shall act as the representative of holders of Company Capital Stock, Qualified Options and Qualified Warrants and shall be authorized to take any action and to make and deliver any certificate, notice, consent or instrument required or permitted to be made or delivered under this Agreement or under the documents referred to in this Agreement by such Persons, to waive any requirements of this Agreement or to enter into one or more amendments or supplements to this Agreement that Stockholders' Representative determines in Stockholders' Representative's sole and absolute discretion to be necessary, appropriate or advisable, which authority includes the execution and delivery of the Escrow Agreement and any amendments or supplements thereto and the performance of all obligations thereunder, including authority to collect and pay funds and dispute, settle, compromise and make all claims. The authority of Stockholders' Representative includes the right to hire or retain, at the sole expense of the holders of Company Capital Stock, Qualified Options and Qualified Warrants, such counsel, investment bankers, accountants, representatives and other professional advisors as Stockholders' Representative determines in Stockholders' Representative sole and absolute discretion to be necessary, appropriate or advisable in order to perform this Agreement and the Escrow Agreement. Notwithstanding the foregoing, the Stockholders' Representative shall not have the authority to amend, on behalf of Company Capital Stock, Qualified Options and Qualified Warrants, (i) the terms of Article II of this Agreement or (ii) any obligation of Company Capital Stock, Qualified Options and Qualified Warrants under this Agreement in a manner that expands such obligation. Any party will have the right to rely upon any action taken by Stockholders' Representative in accordance with the powers set forth above, and to act in accordance with such action without independent investigation.
- (b) Parent will have no liability to any holder of Company Capital Stock, Company Options or Company Warrants or otherwise, in each case arising out of the acts or omissions of Stockholders' Representative, any disputes among holders of Company Capital Stock, Company Options or Company Warrants or with Stockholders' Representative, or, provided that Parent shall have fulfilled its obligations under Sections 2.2(a) and 2.4, the consents described in Sections 7.2(d), 7.2(f), or 7.2(g) hereof. Parent may rely entirely on its dealings with, and notices to and from, Stockholders' Representative and the Company (with copies to the Persons set forth in Section 10.4 hereof) to satisfy any obligations it might have under this Agreement, the Escrow Agreement or any other agreement referred to in this Agreement.

Article X

Miscellaneous

- 10.1 Press Releases and Announcements. Except as otherwise required by law or the rules of any applicable securities exchange, so long as this Agreement is in effect, Parent, Merger Sub and the Company will not, and will not permit any of their respective affiliates or representatives to, issue or cause the publication of any press release or make any other public announcement with respect to the transactions contemplated by this Agreement without the consent of the other parties hereto, which consent shall not be unreasonably withheld or delayed. Parent, Merger Sub and the Company will cooperate with each other in the development and distribution of all press releases and other public announcements with respect to this Agreement and the transactions contemplated hereby, and will furnish the others with drafts of any such releases and announcements as far in advance as possible.
- 10.2 Expenses. Except as otherwise expressly provided for herein, the Company on the one hand, and Parent, on the other hand, will each pay all expenses incurred by each such party in connection with the transactions contemplated hereunder, including legal, accounting, investment banking and consulting fees and expenses incurred in negotiating, executing and delivering this Agreement and the Related Documents (whether the transactions contemplated hereunder are consummated or not). The Company, on the one hand, and Parent, on the other, will each bear 50% of the expenses incurred in connection with the preparation and filing of any filings required by the HSR Act and any other comparable foreign merger or competition laws which are applicable to the transactions contemplated by this Agreement..
- 10.3 <u>Amendment and Waiver</u>. This Agreement may not be amended or waived except in a writing executed by the party against which such amendment or waiver is sought to be enforced. No course of dealing between or among any Persons having any interest in this Agreement will be deemed effective to modify or amend any part of this Agreement or any rights or obligations of any Person under or by reason of this Agreement.
- 10.4 <u>Notices</u>. All notices, demands and other communications to be given or delivered under or by reason of the provisions of this Agreement will be in writing and will be deemed to have been given (i) when delivered if personally delivered by hand (with written confirmation of receipt), (ii) when received if sent by a nationally recognized overnight courier service (receipt requested), (iii) five business days after being mailed, if sent by first class mail, return receipt requested, or (iv) when receipt is acknowledged by an affirmative act of the party receiving notice, if sent by facsimile, telecopy or other electronic transmission device (provided that such an acknowledgement does not include an acknowledgment generated automatically by a facsimile or telecopy machine or other electronic transmission device). Notices, demands and communications to the parties will, unless another address is specified in writing, be sent to the address indicated below:

Notices to Parent:		
ADC Telecommunications, Inc.		
P.O. Box 1101		
Minneapolis, MN 55440-1101		
Attn: Office of the General Counsel		
Fax: (952) 917-0893		
Notices to the Company:		
LGC Wireless, Inc.		
2540 Junction Avenue		
San Jose, CA 95134		
Attn: John G. Niedermaier		
Fax: (408) 952-2630		
Notices to the Stockholders' Representative:		
Robert R. Ackerman, Jr.		
c/o Allegis Capital		
345 Spear Street, Suite 345		
San Francisco, CA 94105		

with a copy to: Dorsey & Whitney LLP 50 South Sixth Street Minneapolis, MN 55402 Attention: Robert A. Rosenbaum Fax: (612) 340-7800 with a copy to: Ropes & Grav LLP One Embarcadero Center, Suite 2200 San Francisco, CA 94111-3711 Attention: Scott D. Elliott Fax: (415) 315-4834 with a copy to: Ropes & Gray LLP One Embarcadero Center, Suite 2200 San Francisco, CA 94111-3711 Attention: Scott D. Elliott

Fax: (415) 278-9794 Fax: (415) 315-4834

10.5 <u>Interpretation</u>. The language used in this Agreement and the other agreements contemplated hereby shall be deemed to be the language chosen by the parties to express their mutual intent, and no rule of strict construction shall be applied against any party. The table of contents and headings of articles and sections herein are for convenience of reference, do not constitute a part of this Agreement, and shall not be deemed to limit or affect any of the provisions hereof. For all purposes of and under this Agreement, the words "include", "includes" and "including" when used herein, shall be deemed in each case to be followed by the words "without limitation." As used in this Agreement, "Person" means any individual, corporation, limited liability company, limited or general partnership, joint venture, association, joint stock company, trust, unincorporated organization or other entity; "knowledge" means the actual knowledge of Ian Sugarbroad, John Niedermaier, Michael Frausing, Daniel Weinblatt and Melissa Madriaga, as such knowledge has been obtained in the normal conduct of the business and including such knowledge as a reasonably prudent Person in such position would have obtained upon the exercise of reasonable diligence (except that the representation of the Company contained in Section 3.14(d)(iii) shall be qualified by the actual knowledge of Persons referred to in such representation); "affiliate" has the meaning assigned to it in Rule 12b-2 of the Securities Exchange Act of 1934; and all amounts shall be deemed to be stated in U.S. dollars, unless specifically indicated otherwise.

- 10.6 No Third Party Beneficiaries. Nothing in this Agreement shall confer any rights upon any Person or entity that is not a party or permitted assignee of a party to this Agreement, *provided*, *however*, that (i) the holders of Company Capital Stock, Company Options and Company Warrants (as to payments to be made to them pursuant to Article II hereof only), and (ii) the officers and directors of the Company (as to Section 6.8 herein only) are each intended third party beneficiaries of this Agreement.
- 10.7 <u>Severability</u>. Whenever possible, each provision of this Agreement will be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Agreement is held to be prohibited by or invalid under applicable law, such provision will be ineffective only to the extent of such prohibition or invalidity, without invalidating the remainder of such provision or the remaining provisions of this Agreement.
- 10.8 <u>Complete Agreement</u>. This Agreement (together with the Exhibits and the Disclosure Schedule and the other documents delivered pursuant hereto or contemplated hereby) contain the complete agreement between the parties and supersede any prior understandings, agreements or representations by or between the parties, written or oral, which may have related to the subject matter hereof in any way, other than the Confidentiality Agreement, which will continue in full force and effect until the Closing and shall survive any termination of this Agreement.
- 10.9 <u>Disclosure Schedule</u>. The Disclosure Schedule contains a series of schedules corresponding to the Sections contained in Article III. Each item so disclosed in the Disclosure Schedule shall constitute an exception to the representations and warranties to which it makes reference and shall be deemed to be disclosed with respect to each section of the Disclosure Schedule to which it relates and/or representation and warranty herein given, without the necessity of repetitive disclosure or cross-reference, so long as such item is fairly described with reasonable particularity and detail and such description provides a reasonable indication that the item applies to another schedule contained in the Disclosure Schedule.
- 10.10 <u>Assignment</u>. This Agreement and all of the provisions hereof will be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, except that neither this Agreement nor any of the rights, interests or obligations hereunder may be assigned by any party hereto without the prior written consent of the other parties hereto.
- 10.11 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, any one of which need not contain the signatures of more than one party, but all such counterparts taken together will constitute one and the same instrument. A facsimile signature will be considered an original signature.

- 10.12 <u>Governing Law</u>. This Agreement shall be governed by, and construed in accordance with, the laws of the State of Delaware regardless of the laws that might otherwise govern under applicable principles of conflicts of law thereof.
- 10.13 <u>Submission to Jurisdiction</u>. Each of parties irrevocably agrees that any legal action or proceeding with respect to this Agreement or for recognition and enforcement of any judgment in respect hereof brought by the other party hereto or its successors or assigns may be brought and determined in the Chancery or other Courts of the State of Delaware or the Federal Courts sitting within the State of Delaware, and each of parties hereby irrevocably submits with regard to any such action or proceeding for itself and in respect to its property, generally and unconditionally, to the exclusive jurisdiction of the aforesaid courts.
- 10.14 <u>Waiver of Jury Trial</u>. EACH OF PARENT, THE COMPANY AND THE STOCKHOLDERS' REPRESENTATIVE HEREBY IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE ACTIONS OF PARENT, THE COMPANY OR THE STOCKHOLDERS' REPRESENTATIVE IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT HEREOF.

[Remainder of page left blank; signature page follows]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement and Plan of Merger as of the day and year first above written.

ADC TELECOMMUNICATIONS, INC.

By:	/s/ Robert E. Switz	
Name:	Robert E. Switz	
Title:	President & CEO	
HAZELTINE ME	RGER SUB, INC.	
By:	/s/ James G. Mathews	
Name:	James G. Mathews	
Title:	President	
LGC WIRELESS	, INC.	
Ву:	/s/ Ian V. Sugarbroad	
Name:	Ian V. Sugarbroad	
Title:	President & CEO	

VOTING AGREEMENT

This **VOTING AGREEMENT**, dated as of October 21, 2007 (the "<u>Agreement"</u>), is by and among ADC Telecommunications, Inc., a Minnesota corporation ("<u>Parent"</u>), and each Stockholder of LGC Wireless, Inc., a Delaware corporation (the "<u>Company</u>"), whose signature is set forth on the signature pages to this Agreement (each a "<u>Stockholder</u>" and, collectively, the "<u>Stockholders</u>"). Capitalized terms which are used but not defined herein shall have the meanings ascribed to them in the Merger Agreement (as defined below).

WITNESSETH:

WHEREAS, concurrently with the delivery of this Agreement, Parent and the Company are entering into an Agreement and Plan of Merger (the "Merger Agreement"), which provides for Hazeltine Merger Sub, Inc., a Delaware corporation and a wholly owned subsidiary of Parent ("Merger Sub"), to be merged with and into the Company in accordance with the Delaware General Corporation Law and the terms of the Merger Agreement, as a result of which the Company will be the surviving corporation and will be a wholly owned subsidiary of Parent (the "Merger");

WHEREAS, the execution of this Agreement by the Stockholders is a condition to Parent's willingness to enter into the Merger Agreement on the terms contained therein;

WHEREAS, the Stockholders own in the aggregate over 50% of the Company Capital Stock issued and outstanding on a fully-diluted, as converted basis; and

WHEREAS, the Stockholders desire that the Company and Parent consummate the Merger contemplated by the Merger Agreement and are willing to enter into this Agreement to induce Parent to enter into the Merger Agreement.

NOW, THEREFORE, in consideration of the premises and of the mutual covenants and agreements set forth herein, the parties hereto, intending to be legally bound, agree as follows:

- 1. Agreement to Vote, Not to Tender. At such time as the Company convenes a meeting of, solicits written consents from or otherwise seeks a vote of, the Company's stockholders for the purpose of considering and approving the Merger and the other transactions contemplated by the Merger Agreement, each of the Stockholders hereby agrees to vote all shares of Company Capital Stock owned or hereafter acquired by such Stockholder as of the time of such meeting or written consent (whether held directly or beneficially) in favor of the Merger and the other transactions contemplated by the Merger Agreement and all other actions necessary or desirable for the consummation of the Merger. In addition, during the term hereof, each of the Stockholders further agrees to vote all such shares against any other transaction presented to the stockholders of the Company that could have the effect of impeding the ability of Parent and the Company to consummate the Merger. Furthermore, during the term hereof, each Stockholder agrees not to tender, sell or otherwise transfer such Stockholder's shares of Company Capital Stock or warrants, options or other rights to acquire Company Capital Stock to any third party.
- 2. IRREVOCABLE PROXY. EACH STOCKHOLDER HEREBY GRANTS TO AND APPOINTS MERGER SUB AND THE PRESIDENT OF MERGER SUB AND THE SECRETARY OF MERGER SUB, IN THEIR RESPECTIVE CAPACITIES AS OFFICERS OF MERGER SUB, AND ANY INDIVIDUAL WHO SHALL HEREAFTER SUCCEED TO ANY SUCH OFFICE OF MERGER SUB, AND ANY OTHER DESIGNEE OF MERGER SUB, EACH OF THEM INDIVIDUALLY, THE STOCKHOLDER'S PROXY AND ATTORNEY-INFACT (WITH FULL POWER OF SUBSTITUTION) TO VOTE OR ACT BY WRITTEN CONSENT WITH RESPECT TO THE STOCKHOLDER'S SHARES OF COMPANY CAPITAL STOCK IN ACCORDANCE WITH SECTION 1 HEREOF. THIS PROXY IS COUPLED WITH AN INTEREST AND SHALL BE IRREVOCABLE

UNTIL THIS AGREEMENT IS TERMINATED PURSUANT TO SECTION 8 HEREOF. THE STOCKHOLDER HEREBY REVOKES ANY PROXY PREVIOUSLY GRANTED BY IT WITH RESPECT TO THE STOCKHOLDER'S SHARES OF COMPANY CAPITAL STOCK. NOTWITHSTANDING ANY OTHER PROVISIONS OF THIS AGREEMENT, THE IRREVOCABLE PROXY GRANTED HEREUNDER SHALL AUTOMATICALLY TERMINATE UPON THE TERMINATION OF THIS AGREEMENT.

- 3. <u>Limitation</u>. Each Stockholder shall retain at all times the right to vote such Stockholder's shares of Company Capital Stock in that stockholder's sole discretion on all matters, other than those set forth in Section 1, that are at any time or from time to time presented for consideration by the Company's Stockholders generally and the proxy granted pursuant to Section 2 hereof shall be inapplicable in such circumstances.
- 4. No Negotiations. Subject to Section 6.5 of the Merger Agreement, from and after the date of the Merger Agreement until the earlier to occur of the Closing Date or termination of the Merger Agreement pursuant to its terms, the undersigned Stockholder shall not directly (i) solicit, initiate or knowingly encourage the submission of any Takeover Proposal from any Person (including any of its or their officers or employees, representatives, agents, or affiliates), other than the Parent or its affiliates or representatives, (ii) participate in any discussions or negotiations regarding, or furnish to any other Person (other than the Parent or its affiliates or representatives) any information with respect to, or otherwise cooperate in any way with, or assist or participate in, facilitate or knowingly encourage any effort or attempt by any other Person to do or seek any of the foregoing or (iii) approve, endorse or recommend any Takeover Transaction (except to the extent specifically permitted pursuant to Section 6.5 of the Merger Agreement), or (iv) enter into any letter of intent or similar document or any contract, agreement or commitment contemplating or otherwise relating to any possible or proposed Takeover Proposal. In addition, from and after the date of this Agreement until the earlier to occur of the Effective Time or termination of the Merger Agreement pursuant to its terms, each Stockholder shall not, directly or indirectly, make or authorize any public statement, recommendation or solicitation in support of any proposal made by any person, entity or group (other than Parent) relating to any of the foregoing.
- 5. <u>Representations, Warranties and Covenants of the Stockholders</u>. The Stockholders severally, but not jointly, hereby represent and warrant to Parent that:
- (a) Each Stockholder has the requisite legal capacity and authority to execute and deliver this Agreement, to perform the obligations of the Stockholder under this Agreement and to consummate the transactions contemplated by this Agreement. If the Stockholder is not an individual, it has taken all required actions to authorize the execution, delivery and performance of this Agreement. This Agreement has been duly executed and delivered by such Stockholder and constitutes a valid and legally binding obligation of such Stockholder enforceable in accordance with its terms, except to the extent that enforceability thereof may be limited by bankruptcy and other similar laws and general principles of equity or public policy;
- (b) Each Stockholder's execution, delivery and performance of this Agreement will not result in the creation of any Lien upon any of the shares of Company Capital Stock held by such Stockholder under any of the terms, conditions or provisions of any contract to which such Stockholder is a party;
- (c) No filing or registration with or notification to and no permit, authorization, consent or approval of, any court, commission, governmental body, regulatory authority, agency or tribunal wherever located is required to be obtained, made or given by any Stockholder in connection with the execution, delivery and performance by any Stockholder of this Agreement; and
- (d) Schedule 1 to this Agreement correctly sets forth the number of shares of Company Capital Stock, options, warrants and any other rights to acquire Company Capital Stock held or owned by each Stockholder as of the date of this Agreement. Each Stockholder has, and will have as of the Effective Time, good title to all of the shares of Company Capital Stock set forth below his, her or its name on the signature page hereto which, as of the Effective

Time, will be free and clear of all liens, security interests and encumbrances or any restrictions on transfer. Each Stockholder understands and agrees that any shares of Company Capital Stock it acquires after the date hereof and before the Effective Time (by exercise of any options, warrants and any other rights to acquire Company Capital Stock) will be subject to the requirements in Section 1 herein.

- 6. <u>Capacity</u>. The parties hereby agree that the Stockholders are executing this Agreement solely in their capacity as Stockholders of the Company. Nothing contained in this Agreement shall limit or otherwise affect the conduct or exercise of any Stockholder's fiduciary duties as an officer or director of the Company.
- 7. <u>Further Assurances</u>. Each Stockholder will, upon the request of Parent, execute and deliver such documents and take such action reasonably requested by Parent to effectuate the purposes of this Agreement and to consummate the transactions contemplated by the Merger Agreement.
- 8. <u>Termination</u>. This Agreement shall terminate upon the earlier of (a) the Effective Time or (b) the termination of the Merger Agreement in accordance with its terms. In the event this Agreement is terminated in accordance with the preceding sentence, this Agreement (and the proxy granted hereby) shall immediately become void, there shall be no liability under this Agreement on the part of Parent, its officers or directors or the Stockholders, and all rights and obligations of the parties to this Agreement shall cease.
- 9. <u>Expenses</u>. Each party hereto shall pay its own expenses incurred in connection with this Agreement, except as otherwise specified in the Merger Agreement.
- 10. Specific Performance. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement, without the necessity of proving damages or posting any bond, and to enforce specifically the terms and provisions hereof in any court of the United States or any state thereof having jurisdiction, this being in addition to any other remedy to which they are entitled at law or in equity.
- 11. Notice. Unless otherwise provided in this Agreement, all notices and other communications under this Agreement shall be in writing and may be given by any of the following methods: (a) personal delivery (with written confirmation of receipt); (b) registered or certified mail, postage prepaid (receipt requested); (c) overnight delivery service (receipt requested); or (d) by facsimile, telecopy or other electronic transmission device (with receipt acknowledgement, provided that such acknowledgement does not include an acknowledgement generated automatically by a facsimile, telecopy or other electronic transmission device). Such notices and communications shall be sent to the appropriate party at its address given below or at such other address for such party as shall be specified by notice given under this Agreement (and shall be deemed given upon receipt by such party or upon actual delivery to the appropriate address):

(a) if to a Stockholder:

To the addresses set forth on Schedule 1 hereto

with a copy to:

Scott D. Elliott, Esq. Ropes & Gray LLP One Embarcadero Center Suite 2200 San Francisco, CA 94111 Fax: (415) 315-4834

(b) if to Parent:

ADC Telecommunications, Inc. P.O. Box 1101 Minneapolis, MN 55440-1101 Attention: Office of the General Counsel Fax: (952) 917-0893

with a copy to:

Dorsey & Whitney LLP 50 South Sixth Street Minneapolis, Minnesota 55402 Attention: Robert A. Rosenbaum Fax: (612) 340-7800

- 12. <u>Parties in Interest</u>. This Agreement shall inure to the benefit of and be binding upon the parties named herein and their respective successors and assigns. Nothing in this Agreement, expressed or implied, is intended to confer upon any person other than the Stockholders, Parent or Merger Sub, or their permitted successors or assigns, any rights or remedies under or by reason of this Agreement.
- 13. Entire Agreement; Amendments. This Agreement, together with the Merger Agreement and the other documents referred to therein, constitutes the entire agreement between the parties with respect to the subject matter hereof and supersedes all prior and contemporaneous agreements and understandings, both written or oral, between the parties with respect to the subject matter hereof. This Agreement may not be changed, amended or modified orally, but only by an agreement in writing signed by the party against whom any waiver, change, amendment, modification or discharge may be sought.
- 14. <u>Assignment</u>. No party to this Agreement may assign any of its rights or delegate any of its obligations under this Agreement (whether by operation of law or otherwise) without the prior written consent of the other parties hereto.
- 15. <u>Interpretation</u>. The language used in this Agreement shall be deemed to be the language chosen by the parties to express their mutual intent, and no rule of strict construction shall be applied against any party. The headings of articles and sections herein are for convenience of reference, do not constitute a part of this Agreement, and shall not be deemed to limit or affect any of the provisions hereof.
- 16. <u>Counterparts</u>. This Agreement may be executed via facsimile in two or more counterparts, each of which, when executed, shall be deemed to be an original and all of which together shall constitute one and the same document.
- 17. <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, without giving effect to the principles of conflicts of laws thereof.
- 18. <u>Validity</u>. The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provisions of this Agreement, which shall remain in full force and effect. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner to the end that transactions contemplated hereby are fulfilled to the extent possible.

19. <u>Stockholders' Representative</u>. Each of the undersigned Stockholders acknowledges the appointment of Robert R. Ackerman, Jr. as the Stockholders' Representative in accordance with the Merger and the Merger Agreement.

[Remainder of page left blank; signature page follows]

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

ADC TELECOMMUNICATIONS, INC.

By: /s/ Robert E. Switz

Print Name: Robert E. Switz Title: President & CEO

Address: 13625 Technology Drive

Eden Prairie, MN 55344

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

ASSOCIATED VENTURE INVESTORS III, L.P.

By: AVI Management Partners, III, L.P.,

Its General Partner /s/ Brian Grossi

Print Name: Brian Grossi Title: General Partner

By:

Address: c/o Shirley Wong

130 Lytton Ave, Suite 210, Palo Alto, CA 94302

cc: 27345 Deer Springs Way, Los Altos Hills, CA 94022

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

AVI CAPITAL, L.P.

By: AVI Capital Management, L.P.,

Its General Partner

By: /s/ Brian Grossi

Print Name: Brian Grossi Title: General Partner

Address: c/o Shirley Wong

130 Lytton Ave, Suite 210, Palo Alto, CA 94302

cc: 27345 Deer Springs Way, Los Altos Hills, CA 94022

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

AVI SILICON VALLEY PARTNERS, L.P.

By: AVI Management Partners, L.P.,

Its General Partner

By: /s/ Brian Grossi

Print Name: Brian Grossi Title: General Partner

Address: c/o Shirley Wong

130 Lytton Ave, Suite 210, Palo Alto, CA 94302

cc: 27345 Deer Springs Way, Los Altos Hills, CA 94022

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

MAYFIELD ASSOCIATES FUND III a California Limited Partnership

By: Mayfield VIII Management, LLC a Delaware Limited Liability Company, Its General Partner

By: <u>/s/ Yogan K. Dalal</u> Title: Managing Member

Address: 2800 Sand Hill Road, Menlo Park, CA 94025

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

MAYFIELD VIII a California Limited Partnership

By: Mayfield VIII Management, LLC A Delaware Limited Liability Company, Its General Partner

By: <u>/s/ Yogan K. Dalal</u> Title: Managing Member

Address: 2800 Sand Hill Road, Menlo Park, CA 94025

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

WIRELESS TRUST

By: /s/ James T. Beck

Its: Administrative Trustee

Name: James T. Beck

Title: Administrative Trustee

Address: 2800 Sand Hill Road, Menlo Park, CA 94025

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

WIRELESS TRUST II

By: /s/ James T. Beck

Its: Administrative Trustee

Name: James T. Beck

Title: Administrative Trustee

Address: 2800 Sand Hill Road, Menlo Park, CA 94025

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

WIRELESS TRUST III

By: /s/ James T. Beck

Its: Administrative Trustee

Name: James T. Beck

Title: Administrative Trustee

Address: 2800 Sand Hill Road, Menlo Park, CA 94025

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

OMERS ADMINISTRATION CORPORATION

(formerly, Ontario Municipal Employees Retirement System)

By: /s/ John W. Macdonald

Print Name: John W. Macdonald

Title: Executive Vice President, Chief Operating Officer

By: <u>/s/ Patrick G. Crowley</u> Print Name: Patrick G. Crowley

Title: Chief Financial Officer

Address: One University Ave, Suite 1200

Toronto, ON M5J2P1, Canada

With a copy to: c/o OMERS Capital Partners Inc.

Royal Bank Plaza, 200 Bay Street, Toronto, Ontario, CANADA M5J2J2, Att: Chantal Thibault Fax: 416-864-3255

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

CRYSTAL INTERNET VENTURE FUND II (BVI), L.P.

By: Crystal Venture II, Ltd.

By: /s/ Daniel Kellogg

Print Name: Daniel Kellogg

Title: V.P.

Address: 1120 Chester Ave; #418, Cleveland, OH 44114

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

CRYSTAL INTERNET VENTURE FUND II (BVI) CRYSTAL VISION, L.P.

By: Crystal Venture II, Ltd.

By: /s/ Daniel Kellogg

Print Name: Daniel Kellogg

Title: V.P.

Address: 1120 Chester Ave; #418, Cleveland, OH 44114

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

MEDIA TECHNOLOGY EQUITY PARTNERS, L.P.

By: /s/ Robert R. Ackerman, Jr.

Print Name: Robert R. Ackerman, Jr.

Title: MANAGING MEMBER OF THE GENERAL PARTNER

Address: 130 Lytton Avenue; Suite 210

Palo Alto, California 94301

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

MEDIA TECHNOLOGY ENTREPRENEURS FUND II, L.P.

By: <u>/s/ Robert R. Ackerman, Jr.</u> Print Name: Robert R. Ackerman, Jr.

Title: Managing member of the general partner

Address: 130 Lytton Avenue; Suite 210

Palo Alto, California 94301

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

FORTUNETECH SEED FUND LTd.

By: <u>/s/ Philip Wang</u> Print Name: Philip Wang

Title: President

Address: 1731 Technology Drive, Suite 760, San Jose, CA 95110

535 Middlefield Rd., Suite 280, Menlo Park, CA 94025

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

FORTUNE TECHNOLOGY FUND I LTD.

By: <u>/s/ Chew Lo-Hou</u> Print Name: Chew Lo-Hou

Title: Chairman

Address: #05-04 Goldbell Towers, 47 Scotts Road, Singapore 228233

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

FORTUNE TECHNOLOGY FUND II LTD.

By: <u>/s/ Chew Low-Hou</u> Print Name: Chew Low-Hou

Title: Chairman

Address: #05-04 Goldbell Towers, 47 Scotts Road, Singapore 228233

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

RANDOLPH ASSOCIATES

By: /s/ Mathew E. Steinmetz

Print Name: Mathew E. Steinmetz

Title: PARTER

Address: 200 E. RANDOLPH DRIVE, CHICAGO, IL 60601

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

WAVELAND TECHNOLOGY PARTNERS, L.P.

By: /s/ Ryan W. Siemens

Print Name: Ryan W. Siemens

Title: MANAGER, WAVELAND VENTURE PARTNERS, LLC HS GP

Address: 19800 MACARTHUR BLVD, STE 650, IRVINE CA 92612

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

IADF CAPITAL PARTNERS LTD.

By: <u>/s/ Jon Goldberg</u>

Print Name: Jon Goldberg

Title: PARTNER

Address: 1436 N. ASTER, CHICAGO, IL 60610

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

FOR AND ON BEHALF OF UOB HERMES ASIA TECHNOLOGY FUND

By: /s/ Kian Wee Seah Name: Kian-Wee Seah

Title: Deputy Managing Director of

UOB Venture Management PTE Ltd

Address: 80 Raffles Place, #30-20, UOB Plaza 2, Singapore 048624, Singapore

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

REMBRANDT VENTURE PARTNERS II, L.P.

By: <u>/s/ Richard Ling</u> Name: Richard Ling Title: Managing Member

Address: 2200 Sand Hill Road, Menlo Park, California 94025

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

ALVARION INC.

By: <u>/s/ Greg Daily</u> Print Name: Greg Daily

Title: President

Address: 2495 Leghorn Street, Mountain View, CA 94043

REVIEWED/APPROVED:

/s/ Lior Barak

Lior Barak, VP Finance

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

By: /s/ Ian Sugarbroad

IAN SUGARBROAD

Address: 2540 Junction Ave, San Jose, CA 95134

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

By: /s/ Mike Frausing

Mike Frausing

Address: 2540 JUNCTION AVE, SAN JOSE CA 95134

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

By: /s/ John O' Connell

John O' connell

Address: 26912 Eleua Rd., Los Altos Hills, Ca 94022

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be executed as of the day and year first written above.

BY: <u>/S/ John Niedermaier</u> JOHN NIEDERMAIER

Address: 2540 JUNCTION AVE, SAN JOSE, CA 95134

news release

For Immediate Release

Contact: Mark Borman - Investor Relations

952.917.0590

mark.borman@adc.com

Jon Secrest - Public Relations 952.917.0232 jon.secrest@adc.com

ADC Announces Agreement to Acquire LGC Wireless

Conference call, October 22, 8:30 a.m. Eastern Time (details below)

Strategic Acquisition of Global Leader in In-Building Wireless Solutions Accelerates Execution of ADC's All IP Radio Access Network (IP RAN)

LGC Wireless Sales Likely to Double in 2007 and Increase ADC's Overall Growth Rate

Combination of ADC and LGC Wireless Product Families Creates an Extensive Set of Market Leading Wireless Capacity and Coverage Options for Our Customers

Further Enhances ADC's Growth Strategy To Become The Global Communications Network Infrastructure Leader By Building Scale In Wireless Connectivity From The Central Office, Through The Outside Plant and Into Customer Premises

LGC Wireless Acquisition More Than Doubles ADC's Wireless Business To Over 9% Of Total ADC Sales

Post-Close Diluted Earnings Per Share From Continuing Operations Expected To Be Non-Dilutive in Fiscal 2008 and Accretive Thereafter, Excluding Acquisition-Related Charges, Intangibles Amortization And Stock Option Expense.

MINNEAPOLIS - October 22, 2007 - ADC (NASDAQ:ADCT, www.adc.com) today announced an agreement to acquire LGC Wireless (www.lgcwireless.com), a market leader in specialized wireless coverage and capacity solutions for carriers and the enterprise. This acquisition represents a central part of ADC's All IP Radio Access Network (IP RAN) strategy announced earlier this year. LGC's network systems mobile switching and wireless office solutions will augment ADC's existing outdoor solutions. The acquisition will further enhance and add scale to ADC's existing line of wireless capacity and coverage solutions in a fast-growing market. It will also support ADC's long-term goal of being the leading global provider of network infrastructure solutions. LGC Wireless had sales of \$83 million in the last 12 months ended September 30, 2007 compared to sales of \$43 million in the year ended December 31, 2006. The company has approximately 240 employees. A presentation describing this strategic acquisition can be obtained at http://www.adc.com/investorrelations/ presentations/.

"LGC Wireless is the acknowledged leader in in-building wireless solutions, which is a strong growth market and is in the early stages of its growth curve. This acquisition fits well with our strategy for capturing the strong growth potential of the wireless capacity and coverage market. Acquiring LGC Wireless further diversifies our revenue base by more than doubling our wireless business to over 9% of total ADC sales and will increase ADC's overall growth rate. This acquisition is expected to be non-dilutive to earnings per share in fiscal 2008 and accretive thereafter, excluding acquisition-related charges, acquired intangibles amortization and stock option expense," said Robert E. Switz, president and CEO of ADC. "The strategic value of this acquisition is created from combining ADC's advanced outdoor wireless solutions, global scale, worldwide customer base, and innovations in connectivity with LGC Wireless' advancements and market leadership in the inbuilding wireless solutions market. The transaction thereby creates a leading platform for serving carriers' capacity and coverage needs."

"ADC has a long history in wireless coverage and capacity solutions with important digital wireless intellectual property. We have been investing in a portfolio of new products built on a vision of the next-generation radio access network," Switz added. "We are building and marketing IP radio access nodes and distributed antenna systems for several important application areas: the home, business and outdoor venues, such as campuses, public sites and hard-to-serve areas like tunnels and canyons. These products are designed to address both coverage and capacity for emerging data and video intensive wireless handsets while providing attractive economics for carriers."

"I am excited at this opportunity to merge LGC Wireless into ADC and strengthen our leadership in the advancement of next-generation wireless coverage and capacity solutions for communications operators and enterprises worldwide," said Ian Sugarbroad, president and CEO of LGC Wireless. "ADC has a strong financial and market position, solid intellectual property and the global reach of sales into more than 130 countries. LGC has market leadership in in-building wireless capacity and coverage solutions. The combined entities will have the resources, distribution and support to make our combined offerings the best available wireless solutions for our customers worldwide."

Summary of Transaction Terms

ADC has agreed to acquire LGC Wireless for a total valuation of approximately \$169 million. This includes a purchase price for the outstanding equity interests of LGC for approximately \$148.5 million. Under the transaction ADC has agreed to pay cash for all the shares of LGC outstanding at the time of the closing. Optionholders of LGC shares will be given the opportunity to either receive a cash payment for their options or an exchange of the options for options to acquire ADC shares. If all of the LGC optionholders elect to exchange their options for options to acquire ADC shares, then ADC would pay approximately \$136 million in cash and issue options to acquire ADC shares worth approximately \$12.5 million. ADC also will assume certain transaction expenses, LGC employee bonuses and the assumption of LGC's current debt, in aggregate approximately \$20.5 million. The transaction is expected to close during the next 30 to 90 days. Closing of the transaction is subject to approval by necessary regulatory authorities as well as customary closing conditions. After closing the transaction, ADC expects to take a charge for various acquisition-related expenses, the amount of which has not been determined. Excluding this charge and future amortization of acquired intangibles and stock option expense, ADC expects the acquisition to be non-dilutive to earnings per share from continuing operations in fiscal 2008 and around \$0.02-0.04 accretive in fiscal 2009.

Exciting Growth Potential

Based on industry analysts' estimates and ADC's own proprietary market research, the wireless capacity and coverage market's global growth potential is significant around the world. This research provides the following estimates:

As many as 70% of cellular phone calls and text messages today are completed indoors. Wireless carriers and enterprise customers alike are looking for ways to provide economical, seamless and dependable solutions inside buildings.

The in-building wireless market is estimated to be \$1.6 billion in 2007 and expected to grow to \$2.9 billion in 2010, a 23% compound growth rate. LGC Wireless is a leading provider of solutions to Fortune 500 companies and is a market leader in the United States, Middle East and Latin America.

Within the wireless in-building coverage and capacity market, active electronics in-building systems are growing at an even faster 33% compound growth rate to address the needs of emerging 3G and 4G networks.

Leading the Advancement of IP RAN Wireless Solutions

ADC's complete line of turn-key broadband and carrier wireless solutions are transforming the wireless network edge, providing coverage and capacity solutions where needed to serve existing 2G, 3G and emerging 4G networks. From our flexible and scalable outdoor radio head platform, FlexWave Universal Radio Head (URH), to extended fiber reach via FlexWave Millimeter Wave (MMW), to indoor and campus coverage and capacity with FlexWave Base Station System (BSS) solutions, to high-speed wireless backhaul with FlexWave WMX WiMAX, carriers consistently look to ADC for reliable solutions. ADC's wireless solutions portfolio also includes the Digivance line of distributed antenna system (DAS) products and the ClearGain Tower-Mounted Amplifier (TMA) and Ground-Mounted Amplifier (GMA) products.

With more than 10,000 systems shipped worldwide to more than 100 countries, LGC Wireless' scalable, high-performance indoor solutions are at work inside buildings in all categories of major venues and industries. In addition, as a leading supplier of compact GSM and CDMA wireless networks, LGC Wireless also provides specialized solutions to mobile operators for a broad range of applications. LGC Wireless' products offer low-cost installation, end-to-end monitoring and management, and easy integration of emerging wireless technologies.

Today's Conference Call at 8:30 a.m. Eastern

ADC will discuss its acquisition of LGC Wireless on a conference call scheduled today, October 22, at 8:30 a.m. Eastern time. The conference call can be accessed by domestic callers at (800) 399-7506 and by international callers at (706) 634-2489. Starting today at 11:30 a.m. Eastern time, the replay of the call can be accessed until 23:59 p.m. Eastern time on October 29, 2007 by domestic callers at (800) 642-1687 and by international callers at (706) 645-9291 (conference ID number is 20265560).

About ADC

ADC provides the connections for wireline, wireless, cable, broadcast, and enterprise networks around the world. ADC's innovative network

infrastructure equipment and professional services enable high-speed Internet, data, video, and voice services to residential, business and mobile subscribers. ADC (NASDAQ: ADCT) has sales into more than 130 countries. Learn more about ADC at www.adc.com.

About LGC Wireless

With more than 10,000 systems shipped worldwide to more than 100 countries, LGC Wireless is the leader in specialized wireless coverage and capacity solutions for carriers and the enterprise. LGC Wireless is based in San Jose, California USA and is privately held. For more information, please visit www.lgcwireless.com.

Cautionary Statement Regarding Forward Looking Information

All forward-looking statements contained herein, particularly those pertaining to ADC's expectations for the performance of the acquisition as well as future operating results, reflect management's current expectations or beliefs as of the date of such statements and are made pursuant to the safe harbor provisions of the Private Securities Litigation Reform Act of 1995. ADC cautions that any forward-looking statements made by us in this report or in other announcements made by us are qualified by important factors that could cause actual results to differ materially from those in the forward-looking statements. These factors include, without limitation: any statements regarding the future performance of the acquisition; our ability to integrate the operations of the acquired businesses with our own operations and to realize planned synergies from the transaction; future sales; profit percentages; earnings per share and other results of operations; expectations or beliefs regarding the marketplace in which we operate and market growth rates related to the acquisition; the sufficiency of our cash balances and cash generated from operating and financing activities for our future liquidity; the demand for equipment by telecommunication service providers, from which a majority of our sales are derived; the fact our business is increasingly dependent on project-based capital deployment initiatives by our customers for which sales are more prone to significant fluctuations; our ability to operate our business to achieve, maintain and grow operating profitability; macroeconomic factors that influence the demand for telecommunications services and the consequent demand for communications equipment; consolidation among our customers, competitors or vendors which could cause disruption in our customer relationships or our displacement as an equipment vendor to the surviving entity in a customer consolidation; our ability to keep pace with rapid technological change in our industry; our ability to make the proper strategic choices with respect to acquisitions or divestitures; increased competition within our industry and increased pricing pressure from our customers; our dependence on relatively few customers for a majority of our sales as well as potential sales growth in market segments we presently feel have the greatest growth potential; fluctuations in our operating results from quarter-to-quarter, which are influenced by many factors outside of our control, such as variations in demand for particular products in our portfolio that have varying profit margins; the impact of regulatory changes on our customers' willingness to make capital expenditures for our equipment and services; financial problems, work interruptions in operations or other difficulties faced by our customers or vendors, which can influence future sales to customers as well as our ability to either collect amounts due us or obtain necessary materials and components; economic and regulatory conditions both in the United States and outside of the United States, as a significant portion of our sales come from non-U.S. jurisdictions; our ability to protect our intellectual property rights and defend against infringement claims made by other parties; possible limitations on our ability to raise additional capital if required, either due to unfavorable market conditions or lack of investor demand; our ability to attract and retain qualified employees in a competitive environment; potential liabilities that could arise if there are design or manufacturing defects with respect to any of our products; our ability to obtain raw materials and components and the prices of those materials and components, which can be subject to volatility; our dependence on contract manufacturers to make certain of our products; changes in interest rates, foreign currency exchange rates and equity securities prices, all of which will impact our results; our ability to successfully defend or satisfactorily settle any pending litigation or litigation that may arise; fluctuations in the telecommunications market, and other risks and uncertainties, including those identified in the section captioned Risk Factors in Item 1A of ADC's Annual Report on Form 10-K/A for the year ended October 31, 2006 and as may be updated in Item 1A of ADC' s subsequent Quarterly Reports on Form 10-Q or other filings we make with the SEC. ADC disclaims any intention or obligation to update or revise any forward-looking statements, whether as a result of new information, future events or otherwise.

-end-